

# Sensor technology with gait as a diagnostic tool for assessment of Parkinson's disease: a survey

Navleen Kour<sup>1</sup> • Sunanda Gupta<sup>1</sup> • Sakshi Arora<sup>1</sup>

Received: 24 July 2020 / Revised: 7 June 2022 / Accepted: 2 July 2022 © The Author(s), under exclusive licence to Springer Science+Business Media, LLC, part of Springer Nature 2022

### Abstract

Parkinson's Disease (PD) is the most precarious chronic disorder of the human brain affecting millions of people globally. In today's world of technical inventions, the diagnosis of PD at its initial stage is a serious issue. The high dependency on the patient's medical history and clinical rating scales have certain limitations as they are very tiring and include recall bias. Therefore, more stable and objective measurements are needed to execute automated and effective detection of PD. The biometric gait has been developed as a reliable diagnostic tool for such a purpose because of its unobtrusiveness. In past years, the applications of sensor devices have been utilized the most to evaluate PD via gait. So, the purpose of this article is to analyze the past and current research toward sensor-based (SB) diagnosis of PD motor symptoms. In this article, we provide a brief description of PD and the related concepts also defining its impact on human gait. This article comprehensively surveys the SB technology and the role of different sensors in PD gait recognition. This study investigates the machine learning paradigms used in PD analysis and their performance evaluation. Several SB PD gait datasets are surveyed and explored considering literature from the last ten years. Also, we exhaustively provide a discussion section in this article to give a clear picture of the results concluded from the analysis of prior sections. At last, this article examines some gaps in the existing studies that need to be addressed and also suggests some measures to tackle such issues using advanced paradigms.

**Keywords** Parkinson's disease · Sensor-based · Gait · Machine learning · Sensor datasets · Clinical gait analysis

Sunanda Gupta Sunanda.gupta@smvdu.ac.in

<sup>&</sup>lt;sup>1</sup> School of Computer Science & Engineering, Shri Mata Vaishno Devi University, Katra 182320, India

# 1 Introduction

The present age of digital scenarios demands more advanced and automated systems to identify the diseased state of a person to cope with the growing rate of pathologies. The evaluation of disease using a few subjective clinical scoring scales doesn't offer true reliability and hence fails to give accurate results. Therefore, to suppress such bottlenecks, the unique attributes of the human body either physiological (e.g. hand geometry, face, iris, etc.) or behavioral (such as speech, gait, etc.) named as biometric, plays an important role in clinical recognition and have shown efficacy over the decades [60, 202]. Among the highlighted traits, gait has been tremendously given more focus of interest by the researchers for clinical analysis because of its exceptional features. Gait simply means the walking pattern of a person and it can be effectively utilized for the purpose due to facilities it provides like being captured at a low resolution [35], easy setup requirements, and difficult to concealed easily [116]. These benefits of gait analysis in combination make it a useful tool to detect abnormal behavior that happened due to any cause involving factors like injury, age, or some disorder [193]. An estimate given by the study in [197] revealed a high burden of disease in deteriorating public quality of life (QOL) [79]. Neurological disorder (ND) can be seen as one of the most predominant causes resulting in great loss of healthy living. The data provided by the study in [183] based on an analysis made on 328 diseases showed a great rise in ND (approx. 5% growth) between 2006 and 2016 and a major reason for most of the deaths. Due to the prediction of higher growth in the number of aged people, this survey article takes PD among all the ND, in concern for its further investigation.

Parkinson's disease (PD), defined by James Parkinson in 1817 [133], is referred to as the most frequent movement disorder graded in second place after Alzheimer's disease. The real causes for the occurrence of PD are undiscovered but a chemical known as dopamine is primarily related to the degraded functioning of the human mid-brain. When a person is usually diagnosed with PD, almost 70% of dopaminergic neurons are already lost and thus most of the symptoms develop gradually. The occurrence of such signs (like slow movements, freezing of gait (FOG), small and fast steps, etc.) provides powerful clues regarding PD stages. One of the main causes behind the incidence of PD involves an increase in age so it can be considered an age-related disorder where the process of aging is unnatural. An incidence reported the effect of PD on different age groups (such as 1:1000 for population over 65 and 1:100 over 75) with its pick-up range of about 60–64 years [30]. In a study involving statistics of PD, it is analyzed that PD has a distinct impact on different countries. The pie chart plotted in Fig. 1 (left) shows the overall prevalence estimates indicating a great rise in the PD-affected



Fig. 1 (Left) Shows overall prevalence estimates of PD growth from 2005 to 2030. (Right) Depicts country-wise projected rise in PD (elaboration of left part) analyzing the gait parameter of an individual is necessary to accurately identify healthy and PD subjects to cure them adroitly

population from 2005 (4.1 million) to 2030 (8.7 million). The graph in part (right) provides a wider view of the left part which demonstrates the country-wise projected growth of PD where the highest rate is seen in China followed by other countries. The data also revealed that PD has more effect on men (about 63%) as compared to women (approx. 37%) [179]. To treat PD, doctors simply rate the disease on available scales and suggest medications to suppress their symptoms. The entire analysis involves the subjective evaluation which can be faulty if not performed by experts. Thus, the development of the automated analysis of gait parameters is necessary to accurately identify healthy as well as PD subjects and cure them adroitly.

#### 1.1 Motivation

In recent years, gait analysis has contributed immensely to the medical field due to its unique usability. Despite having numerous pros, there exist certain cons such as occlusion, clothing, lighting factor, and other issues like use of the costly platform, small population size, heavy pre-processing requirements, etc. that have attracted huge numbers of researchers to provide more robust approaches to PD diagnosis. Working towards this direction, several survey/ review articles have been published by various researchers. Recently, Brognara et al. in [20] performed a survey considering the last ten years of publications to inspect the efficiency and capability of inertial sensors in the evaluation of Spatio-temporal (SPT) gait features of PD subjects. Similarly, another review by Benson et al. in [17] explored various feature extraction techniques useful in PD evaluation and also discussed the role of wearables in PD analysis highlighting their advantages and gaps to be focused on. The summary of some important survey articles until 2019 (year-wise) is given in Table 1.

This survey article systematically goes through many publications in reputed journals and conferences and comes up with approx. 160 such articles that extensively concentrate on sensor data for PD evaluation. The article intends to keenly study PD detection based on gait data captured via sensor devices by exploring the past as well as current literature and also highlights the scope in the future. The fundamental objectives of this survey can be made clear by the points provided:

- The article briefly discusses the various aspects of PD and its relation to human gait considering SB analysis.
- (2) Surveyed SB PD gait evaluation involving more than 150 research articles from reputed journals and conferences.

Ref./Year	Study purpose
[ <mark>20</mark> ]/2019	Focussed last ten years publications to inspect capabilities of IS in evaluation of SPT gait features of PD
[ <mark>126</mark> ]/2019	Reviewed the articles focussing PD FOG analysis via WS, also highlighting the gaps to be overcome in future
[17]/2018	Discussed PD feature extraction techniques and the role of WS in PD analysis
[146]/2018	Ascertained the factors causing PD instability to avoid falling risk of subjects
[201]/2016	Investigated the recent advancements in WS technology in PD motor function evaluation
[122]/2016	Keenly reviewed the WS methods and technology to analyze motion symptoms of PD subjects
[ <mark>30</mark> ]/2013	Focussed PD gait based on WS and advised PD cure strategies

Table 1 Summarizes important survey/review articles (year-wise) on sensor-based PD gait detection

- (3) The article comprehensively investigates sensor modality/technology for PD gait acquisition giving its benefits and drawbacks.
- (4) The article extensively provides the role of artificial intelligence (machine learning paradigms) in the analysis of PD with the help of a percentage usage graph of each as well as their performance analysis.
- (5) The article inclusively surveys and presents an insight into PD SB datasets that can be helpful for further research studies.
- (6) The article exhaustively provides the discussion explaining the outcome obtained from prior sections with proper graphs and charts.
- (7) The article also highlights the gaps in the existing literature and gives possible suggestions to tackle them in the future more efficiently.

#### 1.2 Article organization

After presenting the introduction and key aims of the article in Section 1, the other sections are organized as follows; Section 2 gives a brief description of PD and the associated concepts. The overview of human gait including its phases and gait parameter ranges is provided in Section 3. Section 4 outlines the literature search strategy and its outcomes in the respective sub-sections discussing sensor technology in detail. In Section 5, we investigate various machine learning paradigms (MLP) that have been used for PD diagnosis. Section 6 provides a view of public and private sensor gait datasets. The results obtained from each section are discussed and presented in Section 7, also giving the existing gaps and possible solutions. At last, Section 8 briefly provides the conclusion of the survey article.

### 2 Parkinson's disease (PD)- a progressive ND

Parkinson's disease, widely referred to as chronic degenerative movement disorder, has been identified as the second most precarious neurological disbalance worldwide [15, 127]. This section provides the general scenario in PD as depicted with the help of Fig. 2. and gives a brief description of PD related concepts that include the following.

**PD brain degeneration and etiology** PD generally occurs when cells in a particular part of the brain known as substantia nigra stop working in the usual manner. Due to the release of insufficient dopamine [15], a person starts developing symptoms of PD [36]. Figure 2 clearly shows the deviations in the presence of dopamine amount both in healthy and PD pruned brain that results in improper body functioning. Research related to PD isn't able to identify its actual causes [77, 96] and mainly involves genetic, metabolic, environmental, and other factors. Evidence shows that the presence of environmental toxins such as manganese, carbon monoxide, other pesticides, and herbicides are the major cause of PD. The entire destruction results in faulty mutations in the alpha-synuclein genes which is the key component of Lewy bodies in almost all PD clients.

**PD outcomes/symptoms** The age range of people who are affected by PD is not fixed, but it can affect people of all ages [153, 154]. Two signs either on one or both portions of the body start developing when a person comes under the impact of PD i.e. motor and non-motor



Fig. 2 A pictorial representation of PD with its related concepts (brain damage, causes, symptoms, clinical scales, treatment) [127, 154, 172]. Some of the images used in this and other figures of this article are taken from the internet whose proper URLs are provided in the Appendix

indications [68, 154]. Further, under motor classification, two subcategories are defined i.e. locomotory which happens when a person is moving, and non-locomotory which involves speech problems, and handwriting issues.

**Clinical measures for PD and cure** Till now, there is no specific test for PD diagnosis. Initially, a combination of tests is recommended such as medical resonance imaging (MRI), and positron emission tomography [127]. Then, several scales including the Hoehn and Yahr scale (HNY), Activities of Daily Living (ADL), etc., and some questionnaires are used for evaluation of PD severity [51, 68, 122]. PD being a non-curable brain disorder can be controlled through medications including levodopa, duodopa, dopamine agonists, anticholinergic medicines, etc. [172]. Similarly, some therapies and surgical cures like physiotherapy (to increase muscle flexibility), speech (vocal clarity) and dance therapies, deep brain stimulation, pallidotomy, thalamotomy, etc. are available for PD patients.

# 3 Human gait as a diagnostic tool

Ambulating on two limbs smoothly is the most significant and proficient physical exercise that has attracted huge researchers in the world for pathological analysis. Human gait, being the most decisive biometric came into reflection from the period of Aristotle [12] and demonstrated its importance to the modern era of advanced technology too in several fields [167,

195]. This section gives a brief introduction to the human gait and its crucial need in PD monitoring.

The events of walking appear in the form of a series called gait cycle (GC). The aggregation of two main phases of the gait cycle and its subsequent sub-phases i.e. Stance phase (foot on ground, 60% share in GC- heel-strike, foot-flat, mid-stance, heel-off, and toe-off as sub-phases) and Swing phase (foot in air, 40% share in GC- initial swing, mid-swing, and terminal-swing as sub-phases) in whole completes the total GC which begins from initial contact and further goes on [141]. Every single phase of the GC has its significance and captures three tasks including Weight acceptance (for limb balancing and shock absorption, by the first 2 sub-phases of stance), Single limb support (for load stability, by the next 2 sub-phases of stance), and limb advancement (for body progression, by last stance sub-phase and continues to rest of swing sub-phases) [139, 182].

In the current scenario, the use of gait for clinical monitoring has become a popular research area. Gait analysis can be simply seen as the empiric screening to remark the deformities observable by naked eyes. To truly understand the pathological conditions, it provides a consistent way to analyze the asymmetries while walking [167, 182]. Gait measurements of PD-affected subjects can be performed using different information such as kinetic, kinematic, spatiotemporal, etc. If these parameters are crucially analyzed in-depth, they can provide solid clues for the presence of PD as they differ a lot from the healthier gait. A representation of some important gait parameters is provided in Fig. 3 and Table 2 highlights their respective meaning and their ranges for normal gait.

Firstly, the walking features for normal and abnormal PD subjects are recorded and then compared to note the number of deviations between them. As seen in the table below, there are predefined ranges for every normal gait parameter such as gait speed/velocity (2–3 miles per hour), cadence (80–110 steps per minute), step length (28 in.), etc. So, the alterations (high or low) in a person's gait from the average ranges give indications for the presence of the diseased state. E.g. PD, in which the person has slow walking speed, large cycle time, stooped posture, etc. Therefore, to make an effective diagnosis of gait abnormalities, it is necessary to evaluate at least one GC which is often used by the clinicians for proper treatment of people with such unusual conditions as PD [71, 191].

### 4 Literature search plan

To achieve the basic motive of this article, we performed a comprehensive state-of-art literature search from 2010 to 2020. Several journals of repute such as Springer, sensors, IEEE, Elsevier, Gait & Posture, etc., were explored and simple keywords were used within the



Fig. 3 A depiction of some crucial gait parameters (left) and right- figure detailing the leg flexion and extension)

S.No.	Parameter	Meaning	Normal Range (avg.)
1	Walking speed/ velocity	Distance travelled by a person in the specified duration. The average speed can be defined as the product of cadence and stride length as: Speed=stride length × cadence/120	2–3 mph (60-80 m/min)
2	Cadence	The rate at which a person walks or the no. of steps taken in a given time	80-110 steps/min
3	Step length	Distance covered by a person while taking two steps (one with left foot and other with right). Step length can be defined as: Step length=Distance / no. of steps	(28–30) inches
4	Stride length (Gait Cycle)	Distance between heel strike of one foot to the heel strike of same foot. It comprises of two step lengths i.e. left and right and can be given as: Stride length=2* step length	(56–60) inches
5	Stance/Swing ratio	The ratio of stance period to the swing period	60/40
6	Leg Flexion	The range of leg to flex backward. Flexion movement reduces the angle between two joints	0–60 deg.
7	Leg Extension	The range of leg to extend in forward direction thus increasing the angle between two joints	15 deg.
8	Stride width	Side to side distance between lines of two consecutive foot (generally calculated from the center of ankle joint)	(1-3) inches or (3-8) cm
9	Toe out angle	Made by foot ling axis and forward progression's direction intersection	7 deg.
10	Stride period/time	Time taken from heel strike of one foot to the heel strike of same foot and can be calculated as: Stride time=120/ cadence	1 sec.

Table 2 A list of mostly used gait parameters with their meaning and normal range [71, 191]

search box such as "PD Gait", "Gait based PD analysis", etc., yielding an approximation of (5000–7000) related articles. After refining the search string and following the screening and acceptability benchmark, finally, the most relevant and meaningful articles (160–170) were selected to delve into for further PD analysis.

#### 4.1 Search results/outcomes

Traversing and surveying through the selected articles result in the usage of two broad categories of sensor technology for PD appraisal i.e. wearable and non-wearable. Four main types of sensors under the wearable class seem to be significantly utilized including Inertial, Insole shoe, Clothing, and Electromyography (EMG). Also, Floor sensors have contributed a lot toward non-wearable PD assessment. Similarly, some studies employed the combination of one or two sensors by hybridizing them for better results and accuracy as shown in Fig. 4. To scrutinize the sensor technology in PD monitoring, this sub-section provides a vision towards mostly adopted such sensing devices and various research studies contributing as an important part.

#### 4.1.1 Wearable sensor technology (WST)

Wearable sensors (WS) are devices that are directly integrated into the subject's body to measure motor symptoms [53]. To enable gait monitoring in a daily living environment, the



Fig. 4 Mostly adopted sensor-based devices (wearable and non-wearable) used in PD monitoring

use of wearable sensors fixed to any part (e.g. ankle, foot, leg, arm, etc.) has advanced this research area nowadays [24]. Different types of wearable sensors as presented in Table 3 are adopted to enhance the clinical gait diagnosis.

I. Inertial Sensors (IS)

Inertial sensors abbreviated as Inertial measurement unit (IMU) is the amalgamation of a triaxial accelerometer, gyroscope, and magnetometer where the accelerometer can compute velocity, as well as position, and gyroscopes are rich in calculating angular position [24, 111]. Several researchers used IMUs to capture the gait Spatio-temporal and kinematic features.

The study by Heijmans et al. in [62] developed a new IMU-based system, consisting of an accelerometer with  $\pm 8$  g of amplitude range and a gyroscope having  $\pm 2000$  deg. of range considering 20 subjects with idiopathic PD. The results revealed a high consistency of the new system in monitoring the PD population easily. Another work by Amigo et al. in [104] analyzed the gait variability among PD (22 early, 27 moderate) and 25 healthy groups using an inertial measurement sensor (a tri-axial gyroscope and an accelerometer having a range and resolution of  $\pm 19.62$  ms<sup>2</sup> and 0.00981 m/s<sup>2</sup>). Butt et al. in [22] proposed a system by introducing machine learning-based computer-assisted techniques for PD investigation of 59 participants and two IMUs (one positioned at the foot and the other at hand) were utilized to gather the motion features. To diversify the values among 0 and 1, linear scaling was adopted which can be represented as.

$$V' = \frac{V - \min(V)}{\max(V) - \min(V)} \tag{1}$$

Here, V and V denotes original and normalized feature values. At last, applying a support vector machine (SVM), regression, and neural network (NN) for the purpose yielded the best

Amyotrophic lateral	
Healthy controls (HC), <i>A</i>	Related Studies Ref.
in the table: Number (N), ]	Study Limitations
vcronym of terms used	Conclusion
ised for PD gait collection. A	Study Population Performance
echnology u	Sensor
ies on the sensor t	No. of Sensors
me recent stud	Authors
s Shows sol is (ALS)	
<b>Table 3</b> sclerosi	

No. of Sensors Sensor Study Population Perfor Location (N) One (a tri-axial Thich $\bullet N^{=30}$ $\bullet 0\%$ m	Sensor Study Population Perfor Location (N) Thigh • N=30 • 0% m	Study Population Perfor (N) • N=30 • 0% m	Perfor	mance	Conclusion • Combination of advanced	Study Limitations <ul> <li>Window size limitation</li> </ul>	Related Studies Ref. [5, 9, 10, 13, 14, 25, 27, 37,
oue (a ur-axiat) ingli 7.0-20 accelerometer, (21 PD and 9 HC) error ar gynscope) - <5% on error ar	(21 PD and 9 HC) crost remove (21 PD and 9 HC) crost c	(21 PD and 9 HC) $\sim 50^{\circ}$ m $\sim 50^{\circ}$ m $\sim 50^{\circ}$ m $\sim 50^{\circ}$ m $\sim 50^{\circ}$	error ra • <5% 1 error	te nean rate	AI can improve the efficacy of WS for PD analysis	White ways and a size     Small sample size	10, 2, 10, 13, 14, 22, 27, 27, 42, 42, 47, 52, 55, 63, 64, 70, 74, 76, 82, 86–89, 93, 97, 98, 101,
Three Leg. Thigh •N=10 (PD) • AUC=9 (accelerometer and •Sen=87, esensors) Trunk • Spec.=9	Leg. Thigh • N=10 (PD) • AUC=9 r and • Sen=87 Trunk • Spec.=9	• N=10 (PD) • AUC=9 • Sen=87 • Spec.=9	• AUC=9 • Sen.=87 • Spec.=9	7% % 6%	The use of RNN with LSTM using IS outperforms the existing methods for FOG detection	<ul> <li>Non- evaluation of dependencies among sensors</li> </ul>	103, 108, 110, 117, 119, 124, 130, 131, 134, 137, 142, 145, 147, 155–157, 162, 163, 168, 170, 171, 175, 178, 180, 181,
One (a tri-axial Both Legs • N=8 • Effc.=8K accelerometer (7 PD and 1 HC) • Sen=66 device) • Spec.=8	Both Legs • N=8 • Effc.=8 (7 PD and 1 HC) • Sen=60 • Spec.=8	• N=8 • Eff::=8( (7 PD and 1 HC) • Sen.=60 • Spec.=8	<ul> <li>Effc.=8(</li> <li>Sen.=60</li> <li>Spec.=8</li> </ul>	)% 1.6% 6.6%	<ul> <li>Amalgamation of stimulation and detection provided reliability in PD monitoring</li> </ul>	<ul> <li>Less number of patients data</li> </ul>	184, 185, 187–189, 192, 194, 199, 200, 204, 206]
One (an         Thigh, Waist         N=93 PD (for         AUC=05           accelerometer         LA exp.) and         (for LA)           + gyroscope)         N=38 (for         AUC=0.5           + gyroscope)         FOG study)         (for FOC	Thigh, Waist • N=93 PD (for • AUC=05 LA exp.) and (for LA) N=38 (for • AUC=02 FOG study) (for FOC	• N=93 PD (for • AUC=05 LA exp.) and (for LA) N=38 (for • AUC=05 N=38 (for • AUC=05 FOG study) (for FOC	• AUC=0.5 (for LA) • AUC=0.5 (for FO0	0.2.5	<ul> <li>The use of IMU-based smartphones proved to be helpful to estimate motor fluctuations in PD subjects</li> </ul>	<ul> <li>Less training data</li> <li>Negligence on the disease severity level</li> </ul>	
One (Dynaport/ Fourth and • N=696 (HC) – Opal) Frith Lumbar	Fourth and • N=696 (HC) – Fifth Lumbar	• N=696 (HC) -	I		<ul> <li>Large variability in step time and gait asymmetry seemed to be the best indicators for the presence of PD</li> </ul>	<ul> <li>Consideration of only a single gait model</li> <li>Infeasibility of used statistical models</li> </ul>	
One (a nine-axis         Lower Back         • N=10         • AUC=93.2         inertial         (PD)         • Sens=22.4         accelerometer)         • Spec=94.9         e <td>Lower Back • N=10 • AUC=93.2 (PD) • Sens=92.4 • Spec=94.9</td> <td>• N = 10 • AUC=93.2 (PD) • Sens. = 92.4 • Spec. = 94.9</td> <td><ul> <li>AUC=93.2</li> <li>Sens.=92.4</li> <li>Spec.=94.9</li> </ul></td> <td>% % %</td> <td><ul> <li>The use of sensor and k-means algorithm can successfully determine FOG events automatically</li> </ul></td> <td><ul> <li>Data need to be updated for the designed system due to low sampling frequency</li> </ul></td> <td></td>	Lower Back • N=10 • AUC=93.2 (PD) • Sens=92.4 • Spec=94.9	• N = 10 • AUC=93.2 (PD) • Sens. = 92.4 • Spec. = 94.9	<ul> <li>AUC=93.2</li> <li>Sens.=92.4</li> <li>Spec.=94.9</li> </ul>	% % %	<ul> <li>The use of sensor and k-means algorithm can successfully determine FOG events automatically</li> </ul>	<ul> <li>Data need to be updated for the designed system due to low sampling frequency</li> </ul>	
One (inertial Waist • N=21 • 90% (for G measurement (left side) (PD) between st unit) and spec.) and spec.)	Waist • N=21 • 90% (for G (left side) (PD) between st and spec.)	• N=21 • 90% (for G (PD) between st (PD) and spec.)	• 90% (for G between se and spec.)	M sus	<ul> <li>Deep learning model showed high significance in accurate FOG detection by improving accuracy</li> </ul>	<ul> <li>Computationally expensive for real-time implementation</li> <li>Insufficient sample size</li> </ul>	
One (inertial         Foot         • N=90         • Recall: 96.7           measurement         (30 HC, 30PD and • Acc.= 78%           unit)         30 HJ)         • Spec.= 96.7	Foot • N = 90 • Recall: 96.7 (30 HC, 30PD and • Acc. = 78% 30 IH) • Spec. = 96.7	• N = 90 • Recall: 96.7 (30 HC, 30PD and • Acc. = 78% 30 IH)) • Spec. = 96.7	• Recall: 96.7 • Acc.=78% • Spec.=96.7	%	<ul> <li>The acquired system using machine learning can be helpful in clinical analysis</li> </ul>	<ul> <li>Fewer data samples</li> <li>Non-consideration of upper limb movement</li> </ul>	
One (a tri-axial Waist • N=15 • Sens.=91.79 accelerometer) (PD) • Spec.=87.49 MASPARK	Waist • N=15 • Sens.=91.79 (PD) • Spec.=87.49 MASPARK	• N=15 • Sens.=91.79 (PD) • Spec.=87.46 MASPARK	<ul> <li>Sens.=91.7%</li> <li>Spec.=87.4%</li> </ul>	~ %	• The use of a new FOG detection approach may enhance the true positive rate in PD analysis	• Less sample size	

Authors (Year)	No. of Sensors	Sensor Location	Study Population (N)	Performance	Conclusion	Study Limitations	Related Studies Ref.
Ahn et al. [2]	One (smart glass IS system)	Eyes	• N=10 (PD FOG)	• Acc.=92.86%	<ul> <li>The proposed system can efficiently improve the gait speed as well as step length of PD natients</li> </ul>	• Few sample size	[5, 9, 10, 13, 14, 25, 27, 37, 42, 43, 47, 52, 55, 63, 64, 70, 74, 76, 82, 86–89, 93, 97, 98, 101, 103, 108, 110, 117, 119,
Pham et al. [136]	Three (tri-axial accelerometer)	Shank, Thigh and Lower Back	• N = 34 (10 for development set and 24 for test set)	<ul> <li>Sens.=96%</li> <li>Spec.=79%</li> </ul>	The used     subject-independent     method provided more     accuracy and tolerance	More elaborated techniques for ASD threshold need to be studied for better PD assessment	124, 130, 131, 134, 137, 137, 137, 137, 137, 137, 137, 137
Capecci et al. [92]	One (inertial sensor based smattphone)	Hip joint	• N=20 (PD)	<ul> <li>Sens.=70.1%</li> <li>Spec.=84.1%</li> </ul>	<ul> <li>Study showed the relevance of the Moore-Bachlin algorithm for FOG detection</li> </ul>	Small size of the sample can decrease the Sens. and Spec. values for PD subgroups comparison	194, 199, 200, 204, 206
Nguyen et al. [115]	Two (a three-axis accelerometer and a three-axis gyroscope)	Lateral Side of shoe	• N=119 (PD)	• Increase in AUC up to 19%	<ul> <li>In-depth gait clusters analysis of systematized gait tests is helpful for clinicians to monitor PD</li> </ul>	The used approach required manual labeling work     Constrained number of	[1, 6, 29, 48, 57, 58, 72, 81, 83, 91, 102, 109, 129, 149, 150, 164, 173, 174, 176, 177, 205, 208–210]
Khoury et al. [73]	Sixteen	Both feet	• N=165 (93 PD and 72 HC)	• Avg. Acc.=83%	<ul> <li>The results obtained for features based proposed methodology can help to differentiate PD</li> </ul>	• Use of a few gait features for evaluation	
Zeng et al. [207]	Sixteen	Both feet	• N= 165 (93 PD and 72 HC)	• Acc. Rate: 91.4%- 98.8%	<ul> <li>The proposed combination of PSR, EMD, NN, and ED has a high potential to automatically classify proposed to the proposed of the p</li></ul>	<ul> <li>Limited dataset size</li> <li>Negligence of PD severity level</li> </ul>	
Asuroglu et al. [8]	Sixteen	Both feet	• N=165 (93 PD and 72 HC)	• Acc.=99% • Spec.=99.5%	TD and TC The approach used had the high potential to derive the exact value of PD severity level	Fewer data samples of PD subjects     orbicrity in a dataset in terms     concernent	
Elde et al. [45]	Sixteen	Both feet	• N=64 (15 PD, 13 ALS, 16 HD and 20HC)	• ROC=0.861 • Acc.=90.6%	SVM with RBF shown reliability in differentiating normal and abnormal subjects for DD inconstion	or r D stages • Negligence of PD severity level	
Paragliola et al. [125]	Sixteen	Both feet	• N=64 (15 PD, 13 ALS, 16 HD and 20HC)	• Acc.=90% • Acc. Improvement: 4.28%	<ul> <li>Deep time-series based approach can successfully detect the walking anomalies</li> </ul>	<ul> <li>Time-consuming approach</li> <li>Off-line consideration of classification process</li> </ul>	

Authors (Year)	No. of Sensors	Sensor Location	Study Population (N)	Performance	Conclusion	Study Limitations	Related Studies Ref.
Alam et al. [4]	Sixteen	Both feet	• N=165 (93 PD and 72 HC)	<ul> <li>Acc.=93.6%</li> <li>Sens.=93.1%</li> <li>Spec.=94.1%</li> </ul>	• VGRF data acquired from WS with SVM can be a useful tool for PD	<ul> <li>Investigated only VGRF data</li> </ul>	[1, 6, 29, 48, 57, 58, 72, 81, 83, 91, 102, 109, 129, 149, 150, 164, 173, 174, 176, 177, 205, 208, 2101
Ertugrul et al. [49]	Sixteen	Both feet	• N=165 (93 PD and 72 HC)	• Acc.=87.58% (Highest by Random Forest)	• The proposed Shifted 1D-LBP can be used to analyze complex signal Ann	<ul> <li>Determining optimal</li> <li>P values was difficult as well as time-consuming</li> </ul>	For 7 - 507 (007 (111 (011
Rennie et al. [152]	One (10 m pressure sensor mat-GaitRite)	Under foot	• N=54 (29 PD and 25 HC)	• p<0.0001	<ul> <li>Difference among gait speed and other gait irregularities can estimate the PD and healthy</li> </ul>	<ul> <li>Data collection during an irregular walk</li> <li>A large amount of gait variability</li> </ul>	[34, 50, 107, 132, 151, 158, 198]
Myers et al. [112]	One (GaitRite)	Under foot	• N=37 PD (13 F and 24 NF)	<ul> <li>ROI analysis:-</li> <li>0.01−0.004 and</li> <li>p&lt;=0.0001 (for velocity, step</li> <li>lowth)</li> </ul>	surged as the study suggested the relevance of the BOLD signal in the analysis of PD gait	<ul> <li>Drawback of small sample size</li> <li>Not valid for PD with a severe level</li> </ul>	
Chen et al. [31]	One Und (U-shaped sensing mat)	ler foot	• N=23 (12 PD and 11 HC)	• Acc.=87.1%-95.6%	<ul> <li>The use of SVM based PSO for PD detection seems to be the reliable method</li> </ul>	<ul> <li>Consideration of a few gait features</li> <li>Less sample size</li> </ul>	
Kliener et al. [75]	Two (force Und plates)	ler foot	• N=44 (22 PD and 22 HC)	Positivecorrelation:     90.3%	<ul> <li>COF's pattern can be an efficient tool to analyze the gait deviations among PD mod UC</li> </ul>	• Small sample size of PD	
Young et al. [50]	Two (AMTI force plates)	Under foot	• N=19 PD	<ul> <li>Confidence interval of 95%</li> <li>In PD-FOG, p=0.039,</li> </ul>	and the • Study revealed the capability of sounds as FOG sensing cues	<ul> <li>Spatial conflict issue</li> <li>Consideration of only sound action to study FOG</li> </ul>	
Okuma et al. [118]	One (body-worn sensor)	Full body	• N=36 PD	<ul> <li>Increase in freezing index</li> </ul>	<ul> <li>Study shown significance of wearable sensors for PD FOG detection</li> </ul>	Small PD sample size	I
Niazmand et al. [186]	One (body worn sensor garment)	Full body	• N=6 PD	<ul> <li>Sensiti vity=88.3%</li> <li>Specificity=85.3%</li> </ul>	<ul> <li>The use of sensor-equipped body gament can efficiently detect FOG</li> </ul>	• Very few sample sizes of PD subjects	

Authors (Year)	No. of Sensors	Sensor Location	Study Population (N)	Performance	Conclusion	Study Limitations	Related Studies Ref.
Putri et al. [144]	One	Leg	• N=23 (15 PWP and 8 HC)	• Acc.=88.4%	• The sensor-based analysis of PD can be helpful to evolves its stores	<ul> <li>Severity level of PD was not considered</li> </ul>	1
Lima et al. [38]	Two (a tri-axial accelerometer and a barometer)	Neck	• N=4126 (2063 PD and 2063 HC)	• 95% confidence interval • p<0.0001	<ul> <li>Popport to share a super- PD patients have double falling risk as compared to normal's</li> <li>Study shows the feasibility of the used sensor in daily tro.</li> </ul>	<ul> <li>Self-reported fall events by patients</li> <li>Lack of separate fall detection data</li> </ul>	[23, 39, 40, 100, 106, 114, 160, 211]
Rehman et al. (2019)	Two (a GaitRite mat and a tri-axial accelerometer)	Lower Back, Under foot	• N=196 (93 PD and 103 HC)	•AUC: 87.83 ±7.81%	<ul> <li>Study suggested the relevance of testing protocol choice and gait sensing vystem</li> </ul>	<ul> <li>Only SVM and RF models were compared</li> <li>Consideration of only voting subjects</li> </ul>	
Chan et al. [28]	Two (3 MEMS unit and an AHRS	Upper Limbs (left and right)	• N=38 PD	• Variability: 64%–80%	• The sensor-based analysis of PD can be helpful to explore it efficiently	• Considered only PD subjects • Small sample size	
Jehu et al. [69]	-15 unit) Seven (6 and a force plate)	Both Wrists, Ankles, Lumbar spine, sternum, and	• N=42 (25 PD and 17 HC)	• p<0.05 (fallers vs. non-fallers)	<ul> <li>Study revealed the capability of IS in diagnosing fallers and non-fallers</li> </ul>	- Ounters in scores • Non-consideration of patient's disease severity levels	
Bertoli et al. [18]	Two (2 MIMUs with IS and an instrumented	underfoot Both Ankles, under foot	• N=236 (PD, CI, and HOA)	• Avg. stride mean absolute error=2%	• The sensor-based PD evaluation can be helpful in the treatment of patients	<ul> <li>Only shorter walks of the subjects were instrumented</li> </ul>	
Buckley et al. [21]	Two (3 IMUs and a GaitRite mat)	Lumber and cervical vertebra,	• N=114 (60 PD and 54 OA)	• p<0.05 and d=-0.10	<ul> <li>Study shown significance of IS for PD detection</li> </ul>	Uncontrolled gait speed     No data recording for	
Perumal et al. [135]	17 (16 inertial sensors and a pressure sensor)	Both feet	• N = 165 (93 PD and 72 HC)	• Acc.=86.9% • Sens.=0.72 • Spec.=0.81	<ul> <li>The analysis based on the sensor can be helpful in PD inspection</li> </ul>	<ul> <li>standard augmments</li> <li>Small data samples of PD patients</li> <li>Heterogeneity in dataset among PD stages</li> </ul>	

# 2 Springer

results using SVM with a classification accuracy of 79.66%. Similarly, Prateek et al. in [140] adopted inertial sensors (a 3-axis accelerometer and gyroscope) and have achieved a higher accuracy of the proposed method i.e. 81.03% in determining the true FOG events.

The application of IS for monitoring PD symptoms has been seen in other research work by Piro et al. in [138] using 3D Avatar. Four MARG (Magnetic, Angular Rate, and Gravity) sensors were attached to the wrists and arms of PD as well as normal subjects. The results achieved for Unified Parkinson's Disease Rating Scale (UPDRS = 0.48) and 3D Avatar = 0.47 revealed the effective agreement between both. Another effort by Hao Hu et al. in [65] used the Freezing index (FI) to analyze the FOG frequency between PD and normal gait (with one accelerometer frequency from 0.5 to 3 Hz and another from 3 to 8 Hz).

$$FI(t_m)_{new} = \frac{\int_3^8 |W(t_m, f)|^2 df}{\int_{0.5}^3 |W(t_m, f)|^2 df}$$
(2)

where  $W(t_{m}, f)$  denotes the short-time Fourier transform (STFTM). This study by Hu redefined FI and generalized the spectrogram  $GSPG_x(t_m, f)$  which can be presented as.FI(t).

$$GSPG_x(t_m, f) = STFTM_{wn_1}(t_m, f)^{\alpha_1} \cdot STFTM^*_{wn_2}(t_m, f)^{\alpha_2}$$
(3)

Here,  $\alpha_1$  and  $\alpha_2$  are the STFTM powers and  $wn_1$ ,  $wn_2$  denotes the window size where one is wider and the other is narrower. The experimentation results showed the specificity, sensitivity, and accuracy of 81.83%, 82.66%, and 82.83% respectively. The other work by Yoneyyama et al. in [203] and Gaßner et al. in [56] determined the capability of inertial sensors in future PD detection to help clinicians and for further treatment.

#### II. Shoe Sensors (INS)

Pressure shoe technology, mostly used in clinical research, comprises force-sensitive resistors embedded into foot insoles or fixed directly to the shoe. The key purpose of such a device is to perform spatial and temporal pressure distributions during walking activity [85, 90]. Zeng et al. in [207] proposed a novel methodology to classify the patterns among PD and healthy subjects by utilizing the concept of phase space reconstruction (PSRN), empirical mode decomposition, and NN. Initially, the vertical ground reaction force (VGRF) data was collected from the Physionet dataset (93 PD and 73 normal subjects) using 16 force-resistive insole shoes (8 under each foot). Then using PSRN and empirical decomposition with a neural network for pattern distinction provided the highest accuracy of 81.1%. The use of insole pressure shoe sensors can also be seen in the study by Alafeef et al. in [3] using Physionet. The study aimed to develop a novel wavelet transform-oriented method to identify the subjects with PD by analyzing signals from the mixed frequency-time domain and comes up with an accuracy rate of 97.6% with Artificial NN (ANN).

Ota et al. in [123] aimed to elucidate the connection between some significant PD indicators including stride time fluctuations and PD severity levels. Forty-five PD subjects (24F, 21 M) participated and footstep timing was gathered via footswitch sensors and the collected time series data were evaluated which can be presented as.

$$s(i) = J(i+1) - J(i)$$
 (4)

where s(i) is the i-th stride spell and J(i) means i-th step timing and the procedure shows an accuracy, specificity, and sensitivity of 0.76, 0.86, and 0.93 respectively.

#### III. Clothing Sensors (CS)

Clothing sensor also named the smart fabric is another technology that has contributed immensely to clinical gait evaluation. These devices are equipped with a group of inherent sensors along with cable connections and are capable of measuring entire body movement as well as joint angles. Sensing fabric technology proved to be very robust as it is comfortable and provides better mobility to the user [7]. Okuma et al. by the study in [118] intended to objectively identify the FOG and fall events in subjects with PD in day-to-day activity using a full-body garment. The results obtained by FI in both groups provided reliable detection of FOG as well as fallers.

Similarly, Niazmand et al. in [186] proposed a FOG detection method by adopting a body-worn garment consisting of coherent acceleration to evaluate PD patients. The measurement values fell in the range of  $\pm 2 \text{ gm}_u$  with 0.004 gm<sub>u</sub> resolution and the relative acceleration was computed as.

$$u[l] = a[l] - gm_u \tag{5}$$

$$u[l]' = u[l] - \frac{1}{20} \sum_{l=1}^{l=20} u[l]$$
(6)

$$u[I]' = \left\{ \frac{u[I]', u[I]' > N_{THLDE}}{0, u[I]' \le N_{THLDE}} \right\}$$
(7)

where denotes activity value, u[I] is the relative, and  $gm_u$  is the gravity acceleration, and  $N_{THLDE}$  represents the noise range (max). The new method provided a sensitivity of 88.3% and specificity of 85.3% which indicates the effectiveness of the considered mechanism without any requirement of heavy installations and set-up for FOG inspection.

#### IV. EMG Sensors

In PD, the muscle electrical pattern of a patient differs significantly with disease severity as well as from healthy subjects. Therefore, EMG also named electromyography sensors are the devices employed to act as a diagnostic tool to identify various disorders such as PD [54, 169]. These devices are available in two forms: surface (take measurements directly from the skin surface) and intramuscular (require needle insertion into the skin) EMGs. Former EMG is more user-friendly, less painful, and simpler to use but is restricted only to particular muscles while later can cause discomfort to patients. Putri et al. in the study [144] analyzed the EMG signals using a pattern recognition technique to further differentiate 15 PD and 8 healthy controls. Surface EMG electrodes with a sampling frequency of 100 Hz were placed on three muscles of the subjects. Twelve time-frequency domain features of EMG were considered, as in the example, the integrated EMG (IEMGc) and mean absolute value ( $MAV_l$ ) features were computed as.

$$IEMG_c = \sum_{i=1}^{M} |w_i| \tag{8}$$

$$MAV_{l} = \frac{1}{M} \sum_{i=1}^{M} |w_{i}|$$
(9)

Here, M denotes the total number of patterns and  $w_i$  is a single signal. Using ANN for the purpose achieved an accuracy of 88.4% thus giving the relevance of EMG in PD diagnosis.

#### 4.1.2 Non-wearable sensor technology (NWST)

Contrary to wearables, these types of sensors capture the forces when contact between foot and surface occurs. Non-wearable sensors are non-attachable to the body and make use of well-specialized laboratories having properly marked passages to enable gait recording of a subject. Under these sensors, floor sensor systems have been chiefly adopted to measure ground reaction forces (GRFs) produced during the subject's walk.

#### I Floor Sensors (FS).

Floor Sensors, the technology beneath our feet is the group of several lightweight sensors that are interconnected in the form of the instrumented walkway and are available in several forms like electronic mats, instrumented plates, etc. depending on the type of application [16]. Using such sensors in PD not only allows its detection but can also prevent the falling risk by monitoring the person's footsteps. Kleiner et al. in [75] proposed a study to outline the differences in the coefficient of friction (COFR) among 22 PD and 22 healthy subjects while walking barefoot. They employed two force plates (Kistler, Model 9286BA) and the ratio between shear force and the normal GRF was used to visualize the COFR curve which may be defined as

$$COFR = \frac{(FOY)^2 + (FOX)^2}{FOZ}$$
(10)

Here, FOY, FOX, and FOZ represent the antpost, medio, and VGRF. Using statistical analysis measures provided the significant differences among both the groups based on COFR. In another study, Christofoletti et al. in the study [33] made use of a GaitRite (4.8 m long) instrumented mat to conduct Time up and Go test and a six-minute gait test to identify PD gait impairment factors considering 144 PD subjects. The differences in variance and gait velocities with balance and age showed a positive relationship between these factors and PD mobility. Again, using two computerized force plates (at 100 Hz), Nantel et al. [113] tried to examine the relevance of repetitive stepping in place (SIPL) to identify the freezing events in 30 PD affected subjects as compared to 9 healthy controls. Applying the Spearman rank test yielded the specificity and sensitivity of 93% and 87% thus proving the usefulness of SIPL for the detection of FOG in PD. Results also showed a high correlation with the freezing of gait questionnaire (r = 0.80, p < 0.001).

#### 4.1.3 Merged WST and WST + NWST fusion

Irrespective of using different sensors individually, some researchers adopted the concept of incorporating multiple sensors (either fusing WST or combining WST and NWST) to gain more profitable outcomes. Making an amalgam of sensors not only enhances the system efficiency but provides more reliability as they have very few connection points as well as a low risk of breakage [46]. Mileti et al. [105] tried to monitor the quality of gait for 26 PD patients and 11 healthy subjects in daily activities. The fusion of two IMUs (Xens Technology) and 8 force insole resistors (wireless) was used and the results obtained by measuring the gait phase quality index demonstrated good performance in threshold-based methods (0.25 <  $G_p$  < 0.70) for real-time gait analysis. Similarly, Kugler et al. in [80] proposed a work to perform an objective inspection of PD using the combination of surface EMG (consisting of four electrodes, an amplifier) and an IMU sensor.

Besides merging two same categories of sensors, few authors combined two different categories of sensors i.e. wearable and non-wearable sensor technology [18]. Research by Jehu et al. in [69] studied postural properties in PD by matching and comparing postural

strategies among fallers (n = 11), non-fallers (n = 14), and healthy adults (n = 17). The mixture of six accelerometer sensors (at 128 Hz) and one force plate (Kistler, at 200 Hz) revealed the adoption of varying postural plans among the recruited groups that can provide useful information regarding falling risk. Similarly, the fusion of IMU (with size-4.2\*4.6\*1.5 cm and weight-43 g) and a pressure mat platform (4.27\*0.61 m, at 120 Hz) can be seen in the study conducted by Hundza et al. in [66].

Table 3 provides a brief description of some recent studies purely focused on sensor-based modalities used for PD gait data acquisition including the number of sensors utilized, the accuracy achieved, and the respective drawbacks. The colors used in the second column in the table (i.e. author) denote the different types of sensors preferred under wearable and non-wearable categories for the purpose. Here, the green, blue, pink, orange, sea-green, and purple colors represent the studies that adopted inertial, insole shoe, floor, clothing, EMG, and finally the merged/combined sensors.

# 5 Role of artificial intelligence (AI) in gait based sensor paradigm for PD analysis- a narrow view

This section provides a brief introduction to MLP applied by the researchers in past years (2010–2020), also giving their accuracies in diagnosing PD. There are four main machine learning techniques (MLT):

#### 5.1 Supervised Machine Learning paradigm (SMLP)

SMLP takes labeled datasets and gleans information from them to label new datasets to predict the function. For e.g. If I and O are the given inputs and outputs such as  $I = \{1,2,3,4,5,6,7\}$  and  $O = \{1,4,9,16,25,36,49\}$  then the predicted function will be:-  $O(I)^2$ . Several algorithms fall in this category consisting of SVM, ANN, Adaboost, K-nearest neighbor (KNN), Random Forest (RF), Decision Tree (DT), and Deep Learning (DL).

SVM is a MLT that map input vectors  $(y_j \in \mathbb{R}^M, 1 \le j \le 1)$  to some high dimensional space of features  $\psi(y_j)$  and uses the concept of the hyperplane. Martin et al. in [157] used the SVM classifier and the results obtained showed the highest efficacy of SVM with an accuracy of 99%. ANN [144] is a widely preferred brain-inspired system used by Zeng et al. in [207] (RBF NN) to classify data from the Physionet dataset and gained an accuracy, sensitivity, and specificity of 96.3%, 96.7%, and 95.8%. DT is the tree-based model where each branch corresponds to a possible decision. Lima et al. in [38] studied the association between PD motor fluctuation severities with daily gait and achieved an accuracy of 98.5%. To avoid overfitting issues, RF is one of the best solutions that combine multiple DTs [177]. The concept of Adaboost was practiced by Kim et al. in [74] to classify PD and normal subjects and has shown the best sensitivity of 86%. Toro et al. in [134] performed a study to automatically diagnose the PD using SVM, ANN, and KNN, and accuracy from 86% to 92% was obtained. Apart from classification algorithms, regression models such as linear [192] and logistic [205] have been utilized for PD investigation which involves modeling between a dependent and an independent variable [150, 162].

The decrease in accuracy of the MLT when trained with a large amount of data leads to the development of a more supreme platform known as deep learning (DL) [26, 87] such as recurrent neural networks (RNN), convolutional neural networks, etc. [199]. The use of RNN

can be seen in the study by Torvi et al. in [187] to determine PD FOG events. This platform has advanced the area of ML to a new level but also suffers high computational complexity, lack of transparency, etc.

# 5.2 Unsupervised Machine Learning paradigm (UMLP)

With an unsupervised machine learning paradigm (UMLP), the model is given the unlabeled dataset. In PD research, the k-means algorithm has been used to classify data based on similarity and distinctive nature. Boli et al. in [84] proposed a new wearable system with a K-means algorithm to differentiate FOG and non-FOG events and achieved an accuracy, sensitivity, and specificity of 93.2%, 92.4%, and 94.9% without any need for labeling the training set.

# 5.3 Semi-supervised Machine Learning paradigm (SSMLP)

SSMLP can be seen as the middle ground between supervised and unsupervised learning techniques. One of the examples is the fuzzy rule-based systems that have been also practiced in PD inspection [119]. A novel fuzzy logic approach was utilized by Ciabattoni et al. in [130] to study FOG in PD in which the logic controller is applied to compute the geometric center for the area under the membership function.

# 5.4 Probabilistic Machine Learning paradigm (PMLP)

A type of ML platform that uses the mathematics of probability theory to represent all forms of uncertainty and thereby apply inverse probability is known as PMLP such as Naïve Bayes (NB) and Hidden Markov model (HMM). Ghassemi et al. [57] applied template matching and hierarchical HMMs to perform PD analysis. Two gait tests were conducted wherein 2nd one HMM achieved an F-score of 96%. Similarly, Raykov et al. in [147] used an NB classifier to evaluate the PD and yielded the sensitivity and specificity of 0.85 and 0.97.

# 6 PD sensor datasets description

In PD research, the patients and normal controls data play a crucial job in the diagnosis of disease as well as in the overall performance of the model. There are several forms of data available depending on different data types such as signal, videos, images, text, and fluid thus comprising gait (e.g. Physionet, Daphnet, INIT), speech (Parkinson's speech, oxford PD detection, etc.) [120, 190], handwriting (HandPD, PaHaw (Parkinson's disease *handwriting database*), etc.) [44, 67, 196] and bio-specimen (<sup>18</sup>F-DMFP-PET DS, PPMI (Parkinson's Progression Markers Initiative)) [128, 166] based PD datasets acquired using different criteria including subject count, frequency/sampling rate, access link and other related information as tabulated in Table 4. Since this article is intensively based on the usage of gait biometrics to analyze signal-based data, therefore in this section, we briefly discussed only those PD datasets which are created using the sensor platform. INIT is also the gait-oriented dataset but it contains video data so it is not considered. Finally, there are six such datasets (one self-created and not public and the other five publicly available) used in the aforementioned studies, presented in Fig. 5, and are highlighted in the table with color to cope with the growing rate of PD.

 Table 4
 Summarizes various datasets for PD research emphasizing the focus on sensor-based PD gait datasets (colored). Acronym of terms used in the table: Freezing of gait (FOG), Dataset (DS)

 Dataset (DS)

References	Dataset Type	Dataset (DS) Name	Equipment	Subject's Count	Capt. Condition	Other info.	Access Link
Goldberger et al. [59]		Physionet	Sixteen shoe sensors	93 PD, 73 HC	Straight ground walk for two min.	100 Hz	https://physionet.org/pn3/gaitpdb/
Sinziana et al. [99]		• CUPID Multimodal DS	Nine Inertial sensors	18 PD	Straight and cross walk, 180 and 360 deg. turns	128 Hz	https://joinup.ec.europa.eu/collection/ ehealth/document/cupid-closed- loop-system-personalised-and-home- rehabilitation-people-parkinsons- disease-curid
Bachlin et al. [11]	Gait (sensor) based	• Daphnet FOG DS	Inertial sensor	10 PD	Straight and random walk with 180 and 360 deg. turns	64 Hz	https://archive.ics.uci. edu/ml/datasets/Daphnet+Freezing+ of+Gait
MASPARK [95]		• Maspark DS	Inertial sensor	92 PD	FOG detection activities at patient home such as moving from one room to other	200 Hz	https://fuur.upc.edu/15557508
Hausdorff et al. [61]		<ul> <li>Hausdorff's Gait Dynamics DS</li> </ul>	Floor (Force) sensors	15 PD, 20 HD 16 HC, 30 LS	A walk for five min. in 77 m long path	300 Hz	https://physionet.org/physiobank/ database/gaitndd/
Yoneyama et al. [203]		Self Created	Inertial sensor	12 PD, 11 HC	Level ground walk, spot stepping, asymmeterical walk, jumping actions	100 Hz	1
Ortells et al. [121]	Gait (vision) based	• INIT	A video camera (2D)	10 HC	Lateral view walk	800*400 px	http://www.vision.uji.es/gaitDB
Sakar et al. [161]		<ul> <li>Parkinson's Speech DS</li> </ul>	Trust MC-1500 microphone	20 PD, 20 HC	26 voice samples while speaking sustained vowels	50 Hz- 13 kHz	https://jeeexplore.ieee.org/document/ 6451090
Arroyave et al. [120]	Speech based	<ul> <li>New Spanish Speech Corpus DS</li> </ul>	M-Audio, Fast Track C400	50 PD, 50 HC	Morning time recording in ON-state	SR=96 kbps	http://www.lrec-conf.org/proceedings/ lrec2014/pdf/7_Paper.pdf

References	Dataset Type	Dataset (DS) Name	Equipment	Subject's Count	Capt. Condition	Other info.	Access Link
OPDC [190]		Oxford PD     Detection DS	Smartphone	334 PD, 104 iRBD, 84 HC	Seven tasks to evaluate voice	I	https://archive.ics.uci.edu/ml/datasets/ Parknsons
Weber et al. [196]		• HandPD	Spiral, meander images	74 PD, 12 HC	Six different activities to record 368 samples	I	http://wwwp.fc.unesp.br/~papa/pub/ datasets/Handpd/
Drotar [44]	Handwriting based	• PaHaw	Intuos 4 M tablet	37 PD, 36 HC	Word writing task	200 Hz	https://www.sciencedirect.com/science/ article/pii/S093336571600063?via% 3Dihub
Isenkul et al. [67]		<ul> <li>PD Spiral drawings using digitized graphics tablet DS</li> </ul>	Wacom Cintiq 12WX graphics tablet	62 PD, 15 HC	Static, dynamic and stability handwriting tests	110 Hz-140 Hz	https://archive.ics.uci.edu/ml/datasets/ Parkinson+Disease+Spiral+Drawings +Using+Digitized+Grap
Michael Fox Foundation		IM94 •	VMAT-2 scans	24 PD and HC	Follow-up after 3,6,12 month, MRI tests	I	https://www.ppmi-info.org/access-data- specimens/download-data/
Segovia et al. [166]	Bio-markers based	• 18F-DMFP- PET DS	PET radiotracer	87 PD	Data collection after insertion of radiopharma- ceutical injection	128*128 matric of 2*2 mm voxels	https://www.frontiersin.org/articles/ 10.3389/fninf:2017,00023/full



Fig. 5 Shows Sensor data-based PD Gait Datasets used in the literature

# 6.1 Physionet dataset

Physionet, the most popular PD signal gait DS, was given by Goldberger et al. in 2000 [59]. The dataset contains large data of VGRFs acquired from 93 subjects with PD and 73 HC via 16 shoe sensors (8 sensors each foot). To capture the data, the subjects were asked to perform a 2-minute walk on hurdle-free ground at a natural speed and the output of sensors was then recorded at 100 s/ sec. Along with VGRF, the dataset also includes information about demographic data, PD severity levels, and other important measures. Wu et al. [205] used Physionet to evaluate the deviations among PD and HC and gained an accuracy of 95% using SVM. Similarly, other researchers of ref. [1, 72, 129, 174, 209] studied the gait variations in PD subjects using data from the most preferred Physionet to help clinicians for better PD detection.

# 6.2 CUPID multimodal dataset

CUPID is another sensor-based gait dataset that contains the data of 18 PD patients gathered using IMU. The sensors were attached to both the wrists of the subjects and straight, crosswalks were actioned with 180 and 360-degree turns while passing halls of hospitals and corridors [99]. The basic motive was to evaluate the relationship between hand motion and FOG events during gait. The 24-hour signal data in addition to nine IS was acquired via one galvanic skin sensor, an electrocardiogram, and a spectroscopy sensor. Many authors including [100] used such a setup (i.e. CUPID dataset) to analyze the FOG episodes during the walk and gained about 72% true prediction rate useful for PD investigation.

# 6.3 Daphnet FOG dataset

Daphnet FOG dataset was presented by Bachlin et al. in [11] which acts as a benchmark to evaluate the efficacy of various automated methods in FOG recognition. The signal data was captured using wearable IS at a frequency rate of 64 Hz. Three major tasks were performed by the subject i.e. straight walk, turning walk, and daily living activity task to keep a check on their gait performance. The dataset contains multivariate, time-series data with 237 instances and 9 attributes. The research work by Spyroula et al. in [94] used this DS to capture the acceleration data and to study the FOG effect on PD subjects. The used approach has given an accuracy of 97%. Similarly, ref. [7, 156] utilized Daphnet DS for PD classification to help clinicians.

# 6.4 MASPARK dataset

The collection of data in MASPARK Dataset was performed under the scope of 'freezing in PD: Improving the quality of life with an automatic control system' i.e. MASPARK project at 300 Hz [95]. Almost 92 PD subjects were considered through the H&Y clinical rating scale. The data acquisition was done at the patient's home using wearable IS while performing many activities such as placing an object from one room to another, to represent FOG. Sama et al. in [162] used this dataset including 15 PD subjects with IS worn on their wrist. The evaluation of the subject's gait using Fast Fourier Transform and PCA achieved a sensitivity of 91.7% and specificity of 87.4%.

#### 6.5 Hausdorff's gait dynamic (HGD) dataset

HGD Dataset or Gait in Neurogenerative disease dataset consisted of data collected from 15 PD, 20 Huntington's Disease (HD), 13 lateral sclerosis (LS), and 16 normal subjects. Force resistive sensors were employed at 300 Hz to capture signals during a 5-minute walk trailed on a 77 m long pathway [61]. So, by the force estimates, one can conclude the output which is proportional to the force recorded by these sensors. A proper format was used for the nomenclature of data present such as hunt, park, als and control means patients with HD, PD, LS, and normal subjects followed by a unique number. Proper suffixes were used to represent the file contents (e.g. Suffix 'rit' denotes right foot signal, 'let' means left foot signal, etc.). The dataset also includes demographic data of the subjects and disease severity rated using the H&Y scale. The use of such DS can be seen in a study by Pun et al. in [158] to analyze the SPT of PD subjects. The proposed system achieved an accuracy of 90% respectively.

#### 6.6 Self-created datasets

Apart from the aforementioned publicly available datasets, a lot of researchers performed experimentation and evaluation on their own created datasets according to the requirements of the proposed model. The study by Arora et al. in [5] tried to analyze the gait variations in PD. The data from 10 PD and 10 HC was gathered using IS equipped smartphone and yielded a sensitivity of 96.2% and specificity of 96.9%. In another study by Wu et al. in [198], a U-shaped instrumented walkway was designed using flexible force sensor units. Almost 218 normal and 168 PD subjects were involved and their SPT gait features were extracted. The use of the created setup with RF showed a higher accuracy rate of 92.49%. Similarly, Cole et al. in [34] constructed their signal data by recruiting 10 PD and 2 HC. A 3-axis accelerometer device and an EMG sensor were attached to the subject's body to measure kinetic gait features. The proposed study gained a sensitivity of 83% and specificity of 97%.

#### 7 Discussion

In this article, a thorough analysis of research work from the year 2010 to the current era has been done especially stressing sensor-based gait data for PD detection. Depending on the articles obtained and scrutinized, various findings have been reached that may be beneficial for future PD assessments. Here, in this portion of the article, we have made an effort to discuss such findings for every section presented above, with the help of proper analytical figures.

Section 4 of this article presents the outcome of the literature search performed on articles highlighting various sensors used for PD gait acquisition. Based on this systematic search strategy, we have come up with three major sensor categories i.e. wearable sensors (IS, INS, FS, CS, EMG, Hybrid), non-wearable sensors (FS), and sometimes the fusion of both. Concerning the usage aspect of such modalities, the art-of-literature reveals about 77% of research interest towards WST, 9%

towards NWST (14 articles), and 14% of studies used a combination of both the technologies. The reason behind heavy consideration of WST in this field is due to its more robust properties than NWST which are very expensive and need high maintenance and calibration. Amongst WST, out of a total of 160 research articles, 84 are IS oriented, 36 insole shoe-based, 2 on cloth sensors, and 1 on EMG sensor. Another important factor to consider was the efficacy of each sensor. The performance of a sensor-based gait system depends on various conditions i.e. type of sensor, number of sensors, and its pros and cons. A sensor should be easy to use, implement, and resource-efficient. From the exploration of existing data, it analyzed the IS has shown the highest performance (accuracy = 93%) followed by FS (about 87%) and other sensors. The reason for the superiority of IS observing its usage and high accuracy rate over others may include several advantageous factors such as simple deployment, cost-effectiveness, ease to carry on the body, low power requirements, high flexibility, portability, and better results. Besides, the signal data obtained using such sensors can be effective in solving complex gait problems which are being widely considered in PD research. Considering the combination of sensors, the fusion of IS and INS shows the best precision and recall when placed on the foot, back and leg together. Based on recent publications (2010–2020), the usage and performance comparison of each sensor can be visualized in the graph presented in Fig. 6.

The next part was to decide and select the machine learning paradigm which can provide more performance and reliability.

Section 5 talks about such concerns where various MLPs are looked over to find the most preferred and accurate one among SMLP, UMLP, SSMLP, and PMLP in PD diagnosis. From the literature, it is noticed that supervised learning has been pointed out the most. Considering the usage ratio of different MLTs from recent data, a pie-chart is constructed as presented with the help of Fig. 7. which indicates a high preference for SMLP i.e. 86.8% in which SVM classifiers are widely considered (21.3%) then ANN (16.5%) followed by RF (14%) and others. About 0.7% of UMLP, 1.3% SSLP, and 12.5% of PMLP are seen to be given a share in PD analysis. Also, focusing on the performance of these MLTs, the highest accuracy was shown by SVM (approx. 91%) as shown in Fig. 7. ANN also seemed to provide good results in this concern for PD investigation followed by KNN and other classifiers. The extraordinary characteristics of SVM such as high memory



## Usage and Performance Comparison of PD Gait Acquisition Sensor Modality

Fig. 6 Graphical representation of the percentage usage of different sensors and their performance comparison in PD data acquisition



Fig. 7 Pie-chart representing the usage (left) and performance analysis (right) of various machine learning paradigms (MLP) used in PD analysis

efficiency, kernel trick, ability to deal with unstructured data, regularization features, high stability and performance rate, etc. make it popularly to be used in the classification of PD from normal subjects. ANN's capability to perform parallel processing, deal with incomplete data, high tolerance to a fault and gradual corruption also offer benefits in PD analysis. The best performance values such as accuracy, precision, recall, etc. depend on the type of gait data and the extracted gait parameters. Thus, considering the fusion of analysis techniques, sometimes the combination of SVM and ANN provides the best recall and precision and on the other side, the best values are obtained using the fusion of SVM, KNN, and NB.

The next section i.e. Section 6 discusses another important area that plays a major role in PD concern. The purpose was to find out what are the datasets available for PD, which datasets are publicly present and created by self, how many datasets were principally based on PD sensor gait data, and proceeding with them for further investigation. Thus, the final search concludes with six PD gait sensor datasets of which five are public and others are built by authors and are not public. Therefore, the critical analysis of considered articles from past years highlighted the use of self-created datasets by the authors in great number (almost 75.4%) as compared to others, as revealed by the graph given in Fig. 8. To achieve remarkable performance, the availability of a large amount of data is a necessary prerequisite. But determining the best dataset that fits into the model requirements (e.g. Severity levels data, real-time data, etc.) is sometimes a very challenging task. Therefore, most of the researchers preferred to build and use their self-made data by recruiting the



Fig. 8 The graphical presentation of %age usage of various sensor datasets preferred for PD investigation

PD and HC with proper consent from different areas to fulfill their study aim. However, using such a dataset can lead to a great performance but the unavailability of these datasets publicly is a major drawback. Moving to the publicly available datasets, Physionet has been significantly used (about 18%) due to its wide applications in biomedical research such as it provides free access- to the bulk of VGRF PD and comparable HC data (complex physiological signals), software libraries and also has assistance for its usage procedure. The entire data present has been properly reviewed and thus helping the clinicians to a large extent. Further, about 3% of work is analyzed to be based on Daphnet DS followed by CUPID (approx. 1.8%), MASPARK (almost 1.2%), and HGD (0.6%) datasets. Thus, both the self and publicly created datasets are contributing beneficially towards effective PD assessment. So once the data have been collected, and preprocessed, features are extracted, and then classifiers are used to classify abnormal and normal subjects. The literature has shown that certain parameters such as sensitivity, accuracy, specificity, F-score, etc. have been considered to validate the algorithm performance by comparing them with the others as presented in Table 2. Despite giving great efforts towards this area, several issues have been also confronted by the authors in their studies also highlighted in Table 2 which opens the new scopes for future research. The primary research problem to consider is the small sample size of PD patients and healthy controls i.e. small dataset size. Some researchers only considered PD patients and didn't compare their gait deviation measures with healthy ones while in other research studies; just the healthy subjects were considered just portraying the gait of diseased persons. Next is the issue of not considering PD severity levels into concern. As PD grows slowly and with its progression with time, its symptoms get worsen. So, it is necessary to consider all the severity levels for its early detection. In most of the works, the experiment is conducted only on one severity level of disease i.e. mild, moderate, or severe while ignoring the simultaneous comparison of all the severities. Thirdly is the bottleneck of the high cost of the sensor platform as well as the requirement of a heavy setup. Last is the problem of considering of few gait features for PD evaluation that can result in decreased accuracy of the system. Therefore, future work can be recommended and performed on such mentioned limitations of existing studies by enhancing the quality of the system by using more automated and highly collaborative techniques related to gait for more robust PD analysis. The use of a Vision-based framework (model-based approach) [78] and advanced deep learning for the purpose may be effective due to its high tolerance capability in several scenarios and may offer unique properties such as cost and time effectiveness, easy preprocessing, and background modeling, robust to noisy signal data, etc.

# 8 Conclusion

Parkinson's disease directly affects the functionality of the brain by degenerating dopamine neurons. Among all, gait biometrics can be seen as the effective mode for PD recognition due to its concealed and unobtrusive nature. The existing literature revealed various important aspects that need to be explored for better prediction of PD. Therefore, intending to explore the work and progress on PD via gait, this article comprehensively surveyed ten-year data from 2010 to 2020 focusing on Sensor-based (SB) technology. This survey article briefly summarized the basic concepts behind PD, its relation to a human gait, gait signal acquisition techniques, machine learning approaches used, and various gait datasets preferred over time.

The detailed analysis of state-of-art literature demonstrated a heavy inclination towards sensor technology for PD analysis. The majority of research works made use of wearable sensors in which inertial sensors (IS) have been in great demand due to their high-performance

rate and other benefits including high flexibility. Further, this article extensively investigated the usage of various machine learning paradigms used in PD diagnosis and concluded that supervised MLP including SVM has been universally adopted covering 21.3%, and has achieved the highest performance in PD classification. Extending the previous studies, we surveyed all the gait datasets used in the research for PD inspection. Then the focus is diverted towards the utilization of only gait-based sensor signal DS and found about 75.4% of work surpassingly preferred the creation of self-made PD datasets to achieve the particular criteria of the framework proposed by the authors. Physionet DS, in publicly available datasets, is seemed to be given higher attention i.e. approx. 18% because of its easy accessibility feature.

With the intent to help researchers, academicians, and clinicians to make more effective PD recognition, this survey article provided a discussion to give insight on the outcome achieved in every section to analyze the uncovered areas in state-of-the-art. The analysis of work however shows the highest usage of IS but this technology is prone to certain drawbacks including high maintenance cost, wearing discomfort, mounting issues, heavy chances to get faulty results due to dust or any other blockage, harmful radiation effects, etc. that can cause major damage to a person's health. Therefore, an automated and sensor-free platform such as a vision-based (VB) approach needs to be applied in such analysis to diagnose PD effectively.

Further, the outcome of the survey revealed the non-availability of a robust VB PD gait dataset. This aspect can be focused on and fulfilled by creating cost-effective VB gait DS of PD subjects to perform such experimentation. Concerning MLP, SVM has been practiced the most but the amalgamation of the learning algorithms and use of advanced techniques such as DL, and ensemble learning is not taken into consideration. The diagnostic performance of the system can be improved significantly by combining multiple learning models. Another important analysis indicated that most of the studies lacked the evaluation of PD at different severity and also the comparison of PD subjects to NM gait and other such disorders is not focused. For an in-depth evaluation of a disease, it is worthwhile to measure gait deviations at each severity level. Further, different types of gait features in combination with robust body motion features can be extracted to improve the system's performance. So, the issues in past work performed on PD show several key points that should be focused on to carry out future work in this domain such as evaluating the VB paradigm, assessing PD using kinematic, spatiotemporal, and speeded up robust features despite just considering kinetic data, consideration of disease severities, increasing sample size of PD subject's, comparing multiple diseases also involving NM gait, implementing hybrid ML methods, etc. Thus, the results obtained and the suggestions made in this survey article can help researchers to build and extend the work by covering the highlighted aspects to perform reliable diagnosis and treatment of PD.

### Appendix

The URLs for the images taken from the internet and used in this article are given as under.

Figure 2: http://theconversation.com/from-blood-letting-to-brain-stimulation-200-years-ofparkinsons-disease-treatment-75914 https://www.parkinsonsneurochallenge.org/sarasotaparkinsons-disease-resources/sarasota-what-is-parkinsons-disease.html https://en.wikipedia. org/wiki/Pathophysiology of Parkinson%27s disease

Figure 5: https://www.tdk-electronics.tdk.com/en/2105790/company/press-center/press-releases/press-releases/press-releases/tronics/inertial-sensors%2D%2Dminiaturized-mems-accelerometerwith-excellent-linearity/2105794 https://www.tekscan.com/applications/force-sensitive-insole https://commons.wikimedia.org/wiki/File:AMTI\_OPT464508\_force\_plate.png http://www. mobihealthnews.com/35705/athos-raises-12-2m-for-health-sensing-clothing https://www.aliexpress.com/i/32910049034.html https://www.openhardware.io/view/75/MyMultisensorCoinCell.

Also, all the acronyms used in the paper are given in Appendix Table 5.

Table 5 Glossary

Acronym	Meaning
ADL	Activities of Daily Living
AI	Artificial Intelligence
ANN	Artificial Neural Network
COFR	Coefficient of Friction
DS	Dataset
DT	Decision Tree
DL	Deep Learning
EMG	Electromyography
FI	Freezing Index
FOG	Freezing of Gait
GC	Gait Cycle
HC	Healthy Control
HMM	Hidden Markov Model
HNY	Hoehn and Yahr scale
HD	Huntington's Disease
IMU	Inertial Measurement Unit
IS	Inertial Sensor
KNN	K-nearest neighbor
LS	Lateral Sclerosis
LSTM	Long Short-Term Memory
ML	Machine Learning
MLP	Machine Learning Paradigm
MLT	Machine Learning Technique
MARG	Magnetic, Angular Rate, and Gravity
MRI	Medical Resonance Imaging
NB	Naïve Bayes
NN	Neural Network
ND	Neurological disorder
NWST	Non-Wearable Sensor Technology
PD PD	Parkinson's Disease
PPMI	Parkinson's Progression Markers Initiative
PSRN	Phase Space Reconstruction
PCA	Principal Component Analysis
PMLP	Probabilistic MLP
QOL	Quality of Life
RBF	Radial Basis Function
RF	Random Forest
KNN	Recurrent Neural Network
SSMLP	Semi-Supervised MLP
SB	Sensor-based
SIFI	Short-Time Fourier Transform
5P1	Spatio-tempor
SIL	Stepping in Place
SMLP	Supervised MLP
S V IVI	Support vector Machine
UPDKS	Unified Parkinson's Disease Rating Scale
UMLP	Unsupervised MLP
VUKF	Vertical Ground Reaction Forces
wal	wearable Sensor Lechnology

**Data availability** Data sharing not applicable to this article as no datasets were generated or analysed during the current study.

#### Declarations

Conflict of interest The authors declare that they have no conflict of interest.

### References

- Açıcı K, Erdaş ÇB, Aşuroğlu T, Toprak MK, Erdem H, Oğul H (2017) A random forest method to detect Parkinson's disease via gait analysis. In: International conference on engineering applications of neural networks, pp. 609–619. Springer, Cham. https://doi.org/10.1007/978-3-319-65172-9 51
- Ahn DH, Chung H, Lee H-W, Kang K, Ko P-W, Kim NS, Park T (2017) Smart gait-aid glasses for Parkinson's disease patients. IEEE Trans Biomed Eng 64(10):2394–2402. https://doi.org/10.1109/TBME. 2017.2655344
- Alafeef M, Fraiwan M (2019) On the diagnosis of idiopathic Parkinson's disease using continuous wavelet transform complex plot. J Ambient Intell Humaniz Comput 10(7):2805–2815. https://doi.org/10.1007/ s12652-018-1014-x
- Alam MN, Garg A, Munia TTK, Fazel-Rezai R, Tavakolian K (2017) Vertical ground reaction force marker for Parkinson's disease. PLoS One 12(5). https://doi.org/10.1371/journal.pone.0175951
- Arora S, Venkataraman V, Andong Z, Donohue S, Biglan KM, Ray Dorsey E, Little MA (2015) Detecting and monitoring the symptoms of Parkinson's disease using smartphones: a pilot study. Parkinsonism Relat Disord 21(6):650–653. https://doi.org/10.1016/j.parkreldis.2015.02.026
- Arrazaq GA, Safitri I (2018) VGRF wavelet packet decomposition for Parkinson's disease detection. Int J Appl Eng Res 13(5):2648–2653
- Ascioglu G, Senol Y, Erdem D (2018) A smart garment for gait analysis. J Fashion Technol Textile Eng S4: 004. 3: 2. https://doi.org/10.4172/2329-9568.S4-004
- Aşuroğlu T, Açıcı K, Erdaş ÇB, Toprak MK, Erdem H, Oğul H (2018) Parkinson's disease monitoring from gait analysis via foot-worn sensors. Biocybern Biomed Eng 38(3):760–772. https://doi.org/10.1016/j. bbe.2018.06.002
- Azevedo C, Christine BS, Pissard-Gibollet R, Pasquier M, Espiau B, Geny C (2014) Detection of freezing of gait in Parkinson disease: preliminary results. Sensors 14(4):6819–6827. https://doi.org/10.3390/ s140406819
- Bachlin M, Plotnik M, Roggen D, Maidan I, Hausdorff JM, Giladi N, Troster G (2009) Wearable assistant for Parkinson's disease patients with the freezing of gait symptom. IEEE Trans Inf Technol Biomed 14(2): 436–446. https://doi.org/10.1109/TITB.2009.2036165
- Bachlin M et al (2010) Wearable assistant for Parkinson's disease patients with the freezing of gait symptom. IEEE Trans Inf Technol Biomed 14(2):436–446. https://doi.org/10.1109/TITB.2009.2036165
- Baker R (2007) The history of gait analysis before the advent of modern computers. Gait Posture 26(3): 331–342. https://doi.org/10.1016/j.gaitpost.2006.10.014
- Barth J, Klucken J, Kugler P, Kammerer T, Steidl R, Winkler J, Hornegger J, Eskofier B (2011) Biometric and mobile gait analysis for early diagnosis and therapy monitoring in Parkinson's disease. In: 2011 annual international conference of the IEEE engineering in medicine and biology society, pp. 868-871. IEEE. https://doi.org/10.1109/IEMBS.2011.6090226
- Barth J, Eskofier B, Winkler J, Klucken J (2013) Individualized rating of motor impairment using sensorbased gait analysis in Parkinson's disease by multiparametric regression. Basal Ganglia 1(3):52–53. https:// doi.org/10.1016/j.baga.2013.01.037
- Bene R, Antic S, Budisic M, Lisak M, Trkanjec Z, Demarin V, Podobnik-Sarkanii S (2009) Parkinson's Disease. Acta Clin Croat 48(3):377–380. https://hrcak.srce.hr/45288. Accessed 21 Apr 2020
- Benedette (2018) AZO Sensors: The Rise of Floor Sensors (online). https://www.azosensors.com/article. aspx?ArticleID=1006. Accessed 9 Mar 2020
- Benson LC, Clermont CA, Bošnjak E, Ferber R (2018) The use of wearable devices for walking and running gait analysis outside of the lab: a systematic review. Gait Posture 63:124–138. https://doi.org/10. 1016/j.gaitpost.2018.04.047
- Bertoli M, Cereatti A, Trojaniello D, Avanzino L, Pelosin E, Del Din S, Rochester L et al (2018) Estimation of spatio-temporal parameters of gait from magneto-inertial measurement units: multicenter

validation among Parkinson, mildly cognitively impaired and healthy older adults. Biomed Eng Online 17(1):58. https://doi.org/10.1186/s12938-018-0488-2

- Borzi L, Varrecchia M, Olmo G, Artusi CA, Fabbri M, Rizzone MG, Romagnolo A, Zibetti M, Lopiano L (2019) Home monitoring of motor fluctuations in Parkinson's disease patients. J Reliab Intell Environ 5(3):145–162. https://doi.org/10.1007/s40860-019-00086-x
- Brognara L, Palumbo P, Grimm B, Palmerini L (2019) Assessing gait in Parkinson's disease using wearable motion sensors: a systematic review. Diseases 7(1):18. https://doi.org/10.3390/diseases7010018
- Buckley C, Galna B, Rochester L, Mazzà C (2017) Quantification of upper body movements during gait in older adults and in those with Parkinson's disease: impact of acceleration realignment methodologies. Gait Posture 52:265–271. https://doi.org/10.1016/j.gaitpost.2016.11.047
- Butt AH, Rovini E, Esposito D, Rossi G, Maremmani C, Cavallo F (2017) Biomechanical parameter assessment for classification of Parkinson's disease on clinical scale. Int J Distrib Sensor Netw 13(5). https://doi.org/10.1177/1550147717707417
- Caetano MJD, Lord SR, Allen NE, Brodie MA, Song J, Paul SS, Canning CG, Menant JC (2018) Stepping reaction time and gait adaptability are significantly impaired in people with Parkinson's disease: implications for fall risk. Parkinsonism Relat Disord 47:32–38. https://doi.org/10.1016/j.parkreldis.2017.11.340
- Camille J, Shanahan FB, Lizama LEC, Strik M, Moffat BA, Khan F, Kilpatrick TJ, van WegenDer Walt AEEH, Galea MP, Kolbe SC (2018) Technologies for advanced gait and balance assessments in people with multiple sclerosis. Front Neurol 8:708. https://doi.org/10.3389/fneur.2017.00708
- Campbell Dewey D, Miocinovic S, Bernstein I, Khemani P, Dewey RB III, Querry R, Chitnis S, Dewey RB Jr (2014) Automated gait and balance parameters diagnose and correlate with severity in Parkinson disease. J Neurol Sci 345(1–2):131–138. https://doi.org/10.1016/j.jns.2014.07.026
- Camps J, Sama A, Martin M, Rodriguez-Martin D, Perez-Lopez C, Moreno JM, Arostegui JC et al (2018) Deep learning for freezing of gait detection in Parkinson's disease patients in their homes using a waistworn inertial measurement unit. Knowl-Based Syst 139:119–131. https://doi.org/10.1016/j.knosys.2017. 10.017
- Caramia C, Torricelli D, Schmid M, Muñoz-Gonzalez A, Gonzalez-Vargas J, Grandas F, Pons JL (2018) IMU-based classification of Parkinson's disease from gait: a sensitivity analysis on sensor location and feature selection. IEEE J Biomed Health Inf 22(6):1765–1774. https://doi.org/10.1109/JBHI.2018. 2865218
- Chan PY, Ripin ZM, Halim SA, Tharakan J, Muzaimi M, Ng KS, Kamarudin MI et al (2017) An inlaboratory validity and reliability tested system for quantifying hand-arm tremor in motions. IEEE Trans Neural Syst Rehab Eng 26(2):460–467. https://doi.org/10.1109/TNSRE.2017.2782361
- 29. Chang D, Alban-Hidalgo M, Hsu K (2014) Diagnosing Parkinson's disease from gait. In Stanford
- Chen P-H, Wang R-L, Liou D-J, Shaw J-S (2013) Gait disorders in Parkinson's disease: assessment and management. Int J Gerontol 7(4):189–193. https://doi.org/10.1016/j.ijge.2013.03.005
- Chen X, Yao X, Tang C, Sun Y, Wang X, Wu X (2018) Detecting Parkinson's disease using gait analysis with particle swarm optimization. In: International conference on human aspects of IT for the aged population, pp. 263–275. Springer, Cham. https://doi.org/10.1007/978-3-319-92037-5\_20
- Chomiak T, Xian W, Pei Z, Hu B (2019) A novel single-sensor-based method for the detection of gaitcycle breakdown and freezing of gait in Parkinson's disease. J Neural Transm 126(8):1029–1036. https:// doi.org/10.1007/s00702-019-02020-0
- Christofoletti G, McNeely ME, Campbell MC, Duncan RP, Earhart GM (2016) Investigation of factors impacting mobility and gait in Parkinson disease. Hum Mov Sci 49:308–314. https://doi.org/10.1016/j. humov.2016.08.007
- Cole U, Gu H, Dong Z, Artan NS (2016) Classification and visualization tool for gait analysis of Parkinson's disease. In: 2016 38th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), pp. 2407–2410. IEEE, 2016. https://doi.org/10.1109/EMBC. 2016.7591215
- Daehee K, Paik J (2010) Gait recognition using active shape model and motion prediction. IET Comput Vis 4(1):25–36. https://doi.org/10.1049/iet-cvi.2009.0009
- Davie CA (2008) A review of Parkinson's disease. Br Med Bull 86(1):109–127. https://doi.org/10.1093/ bmb/ldn013
- de Lima ALS, Evers LJW, Hahn T, De Vries NM, Daeschler M, Boroojerdi B, Terricabras D, Little MA, Bloem BR, Faber MJ (2018) Impact of motor fluctuations on real-life gait in Parkinson's patients. Gait Posture 62:388–394. https://doi.org/10.1016/j.gaitpost.2018.03.045
- de Lima ALS, Smits T, Darweesh SKL, Valenti G, Milosevic M, Pijl M, Baldus H, de Vries NM, Meinders MJ, Bloem BR (2020) Home-based monitoring of falls using wearable sensors in Parkinson's disease. Mov Disord 35(1):109–115. https://doi.org/10.1002/mds.27830

- De Venuto D, Annese VF, Mezzina G, Defazio G (2018) FPGA-based embedded cyber-physical platform to assess gait and postural stability in Parkinson's disease. IEEE Trans Compon Packag Manuf Technol 8(7):1167–1179. https://doi.org/10.1109/TCPMT.2018.2810103
- Del Din S, Godfrey A, Rochester L (2015) Validation of an accelerometer to quantify a comprehensive battery of gait characteristics in healthy older adults and Parkinson's disease: toward clinical and at home use. IEEE J Biomed Health Inf 20(3):838–847. https://doi.org/10.1109/JBHI.2015.2419317
- Del Din S, Elshehabi M, Galna B, Hobert MA, Warmerdam E, Suenkel U, Brockmann K et al (2019) Gait analysis with wearables predicts conversion to parkinson disease. Ann Neurol 86(3):357–367. https://doi. org/10.1002/ana.25548
- Delval A, Tard C, Rambour M, Luc D, Moreau C (2015) Characterization and quantification of freezing of gait in Parkinson's disease: can detection algorithms replace clinical expert opinion? Neurophysiol Clin/ Clin Neurophysiol 45(4–5):305–313. https://doi.org/10.1016/j.neucli.2015.09.009
- Dijkstra B, Kamsma Y, Zijlstra W (2010) Detection of gait and postures using a miniaturised triaxial accelerometer-based system: accuracy in community-dwelling older adults. Age Ageing 39(2):259–262. https://doi.org/10.1093/ageing/afp249
- Drotár P, Mekyska J, Rektorová I, Masarová L, Smékal Z, Faundez-Zanuy M (2016) Evaluation of handwriting kinematics and pressure for differential diagnosis of Parkinson's disease. Artif Intell Med 67: 39–46. https://doi.org/10.1016/j.artmed.2016.01.004
- Elden RH, Ghoneim VF, Al-Atabany W (2018) A computer aided diagnosis system for the early detection of neurodegenerative diseases using linear and non-linear analysis. In: IEEE 4th Middle East Conference on Biomedical Engineering (MECBME), pp. 116–121. https://doi.org/10.1109/MECBME.2018.8402417
- Electronic Products (2013) Designing in Combination sensors (online). https://www.electronicproducts. com/Sensors\_and\_Transducers/Sensors/Designing\_in\_combination\_sensors.aspx#. Accessed 15 Jan 2020
- Ellis RJ, Ng YS, Zhu S, Tan DM, Anderson B, Schlaug G, Wang Y (2015) A validated smartphone-based assessment of gait and gait variability in Parkinson's disease. PLoS One 10(10). https://doi.org/10.1371/ journal.pone.0141694
- Enas A, Arunkumar N, Narasimhan K, Elamaran V, Venkatraman V (2018) Gait and tremor investigation using machine learning techniques for the diagnosis of Parkinson disease. Futur Gener Comput Syst 83: 366–373. https://doi.org/10.1016/j.future.2018.02.009
- Ertuğrul ÖF, Kaya Y, Tekin R, Almalı MN (2016) Detection of Parkinson's disease by shifted one dimensional local binary patterns from gait. Expert Syst Appl 56:156–163. https://doi.org/10.1016/j.eswa. 2016.03.018
- Eun HY, Hwang S, Kim K, Chung W-H, Youn J, Cho JW (2016) Postural sensory correlates of freezing of gait in Parkinson's disease. Parkinsonism Relat Disord 25:72–77. https://doi.org/10.1016/j.parkreldis.2016. 02.004
- European Parkinson's Disease Association (2017) About Parkinson's (online). https://www.epda.eu.com/ about-parkinsons/symptoms/rating-scales/. Accessed 10 Nov 2019
- Exarchos TP, Tzallas AT, Baga D, Chaloglou D, Fotiadis DI, Tsouli S, Diakou M, Konitsiotis S (2012) Using partial decision trees to predict Parkinson's symptoms: a new approach for diagnosis and therapy in patients suffering from Parkinson's disease. Comput Biol Med 42(2):195–204. https://doi.org/10.1016/j. compbiomed.2011.11.008
- Fang F, Aabith S, Homer-Vanniasinkam S, Tiwari MK (2017) High-resolution 3D prin ting for healthcare underpinned by small-scale fluidics. 3D Printing in Medicine, Woodhead Publishing, 167–206. https://doi. org/10.1016/B978-0-08-100717-4.00023-5
- Farnsworth B (2018) IMOTIONS: What is EMG and how does it work (online). https://imotions.com/ blog/electromyography-101/. Accessed 15 Dec 2019
- Fietzek UM, Stuhlinger L, Plate A, Ceballos-Baumann A, Bötzel K (2017) Spatial constraints evoke increased number of steps during turning in Parkinson's disease. Clin Neurophysiol 128(10):1954–1960. https://doi.org/10.1016/j.clinph.2017.07.399
- Gaßner H, Raccagni C, Eskofier BM, Klucken J, Wenning GK (2019) The diagnostic scope of sensorbased gait analysis in atypical parkinsonism: further observations. Front Neurol 10:5. https://doi.org/10. 3389/fneur.2019.00005
- Ghassem NH, Hannink J, Martindale CF, Gaßner H, Müller M, Klucken J, Eskofier BM (2018) Segmentation of gait sequences in sensor-based movement analysis: a comparison of methods in Parkinson's disease. Sensors 18(1):145. https://doi.org/10.3390/s18010145
- Godinho C, Ferret-Sena V, Brito J, Melo F, Silva Dias M (2016) Postural behavior and Parkinson's disease severity. In: 2016 1st International Conference on Technology and Innovation in Sports, Health and Wellbeing (TISHW), pp. 1–6. IEEE, 2016. https://doi.org/10.1109/TISHW.2016.7847783

- Goldberger A, Amaral L, Glass L, Hausdorff J, Ivanov PC, Mark R, Mietus JE, Moody GB, Peng CK, Stanley HE (2000) PhysioBank, PhysioToolkit, and PhysioNet: components of a new research resource for complex physiologic signals. Circulation 101(23):e215–e220
- Gupta D, Sundaram S, Khanna A, Hassanien AE, De Albuquerque VHC (2018) Improved diagnosis of Parkinson's disease using optimized crow search algorithm. Comput Electr Eng 68:412–424. https://doi. org/10.1016/j.compeleceng.2018.04.014
- Hausdorff JM, Mitchell SL, Firtion R, Peng C-K, Cudkowicz ME, Wei JY, Goldberger AL (1997) Altered fractal dynamics of gait: reduced strideinterval correlations with aging and Huntington's disease. J Appl Physiol 82(1):262–269
- Heijmans M, Habets JGV, Herff C, Aarts J, Stevens A, Kuijf ML, Kubben PL (2019) Monitoring Parkinson's disease symptoms during daily life: a feasibility study. NPJ Parkinson's Dis 5(1):1–6. https://doi.org/10.1038/s41531-019-0093-5
- Hobert MA, Nussbaum S, Heger T, Berg D, Maetzler W, Heinzel S (2019) Progressive gait deficits in Parkinson's disease: a wearable-based biannual 5-year prospective study. Front Aging Neurosci 11:22. https://doi.org/10.3389/fnagi.2019.00022
- Hori K, Hirobe Y, Orimo S, Sawada H, Inaba A, Miyake Y (2017) Early detection of Parkinson's disease based on gait trajectory analysis using wearable sensors. J Neurol Sci 381:348. https://doi.org/10.1016/j. jns.2017.08.989
- Hu H, Ding J-J, Lin K-H, Yang W-C (2014) Freezing of gaits detection for Parkinson's disease patients using fast time-frequency analysis methods and onset detection. In: 2014 IEEE International Conference on Consumer Electronics-Taiwan, pp. 191–192.
- Hundza SR, Hook WR, Harris CR, Mahajan SV, Leslie PA, Spani CA, Spalteholz LG, Birch BJ, Commandeur DT, Livingston NJ (2013) Accurate and reliable gait cycle detection in Parkinson's disease. IEEE Trans Neural Syst Rehab Eng 22(1):127–137. https://doi.org/10.1109/TNSRE.2013.2282080
- Isenkul M, Sakar B, Kursun O (2014) Improved spiral test using digitized graphics tablet for monitoring Parkinson's disease. In: Proc of the Int'l Conf on e-Health and Telemedicine. 171–5
- Jankovic J (2008) Parkinson's disease: clinical features and diagnosis. J Neurol Neurosurg Psychiatry 79(4):368–376. https://doi.org/10.1136/jnnp.2007.131045
- Jehu D, Nantel J (2018) Fallers with Parkinson's disease exhibit restrictive trunk control during walking. Gait Posture 65:246–250. https://doi.org/10.1016/j.gaitpost.2018.07.181
- Jochen K, Barth J, Kugler P, Schlachetzki J, Henze T, Marxreiter F, Kohl Z et al (2013) Unbiased and mobile gait analysis detects motor impairment in Parkinson's disease. PLoS One 8(2). https://doi.org/10. 1371/journal.pone.0056956
- Kharb A, Vipin S, Jain YK, Dhiman S (2011) A review of gait cycle and its parameters. Int J Comput Eng Manag 13:78–83
- Khorasani A, Daliri MR (2014) HMM for classification of Parkinson's disease based on the raw gait data. J Med Syst 38(12):147. https://doi.org/10.1007/s10916-014-0147-5
- Khoury N, Attal F, Amirat Y, Oukhellou L, Mohammed S (2019) Data-driven based approach to aid parkinson's disease diagnosis. Sensors 19(2):242. https://doi.org/10.3390/s19020242
- 74. Kim H, Lee HJ, Lee W, Kwon S, Kim SK, Jeon HS, Park H et al. (2015) Unconstrained detection of freezing of Gait in Parkinson's disease patients using smartphone. In: 2015 37th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), pp. 3751–3754. https:// doi.org/10.1109/EMBC.2015.7319209
- 75. Kleiner A, Galli M, Franceschini M, De Pandis MF, Stocchi F, Albertini G, de Barros RML (2017) The coefficient of friction in Parkinson's disease gait. Funct Neurol 32(1):17
- Koop MM, Ozinga SJ, Rosenfeldt AB, Alberts JL (2018) Quantifying turning behavior and gait in Parkinson's disease using mobile technology. IBRO reports 5:10–16. https://doi.org/10.1016/j.ibror. 2018.06.002
- Kouli A, Torsney KM, Kuan W-L (2018) Parkinson's disease: etiology, neuropathology, and pathogenesis. Parkinson's disease: pathogenesis and clinical aspects [internet]. Codon Publications https://doi.org/ 10.15586/codonpublications.parkinsonsdisease.2018.ch1
- Kour N, Sunanda SA (2019) Computer-vision based diagnosis of Parkinson's disease via gait: a survey. IEEE Access 7:156620–156645. https://doi.org/10.1109/ACCESS.2019.2949744
- Kour N, Gupta S, Arora S (2020) A survey of knee osteoarthritis assessment based on gait. Arch Comput Methods Eng:1–41. https://doi.org/10.1007/s11831-019-09379-z
- Kugler P, Jaremenko C, Schlachetzki J, Winkler J, Klucken J, Eskofier B (2013) Automatic recognition of Parkinson's disease using surface electromyography during standardized gait tests. In: 2013 35th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), pp. 5781– 5784. https://doi.org/10.1109/EMBC.2013.6610865

- Kwon Y, Park SH, Kim J-W, Ho Y, Jeon H-M, Bang M-J, Jung G-I et al (2014) A practical method for the detection of freezing of gait in patients with Parkinson's disease. Clin Interv Aging 9:1709. https://doi.org/ 10.2147/CIA.S69773
- Lan K-C, Shih W-Y (2015) Early detection of neurological disease using a smartphone: a case study. In: 2015 9th international conference on sensing technology (ICST), pp. 461-467, IEEE. https://doi.org/10. 1109/ICSensT.2015.7438443
- Lee S-H, Lim JS (2012) Parkinson's disease classification using gait characteristics and wavelet-based feature extraction. Expert Syst Appl 39(8):7338–7344. https://doi.org/10.1016/j.eswa.2012.01.084
- Li B, Zhang U, Tang L, Gao C, Gu D (2018) Automatic detection system for freezing of gait in Parkinson's Disease based on the clustering algorithm. In: 2nd IEEE Advanced Information Management, Communicates, Electronic and Automation Control Conference (IMCEC), pp. 1640–9. 978-1-5386-1803-5/18
- Lin Shu, Tao Hua, Yangyong Wang, Qiao Li, David Dagan Feng, Xiaoming Tao (2010) In-shoe plantar pressure measurement and analysis system based on fabric pressure sensing array. IEEE Trans Inf Technol Biomed 14(3): 767–775. https://doi.org/10.1109/TITB .2009.2038904
- Lin W-Y, Chou W-Z, Wu W-S, Lee M-Y (2016) An accelerometer-based festinating gait detection algorithm and its application to Parkinson's Disease development. In: 2016 IEEE International Conference on Systems, Man, and Cybernetics (SMC), pp. 003577–003580. https://doi.org/10.1109/ SMC.2016.7844788
- Lonini L, Dai A, Shawen N, Simuni T, Poon C, Shimanovich L, Daeschler M, Ghaffari R, Rogers JA, Jayaraman A (2018) Wearable sensors for Parkinson's disease: which data are worth collecting for training symptom detection models. NPJ Digit Med 1(1):1–8. https://doi.org/10.1038/s41746-018-0071-z
- Lorenzi P, Rao R, Romano G, Kita A, Serpa M, Filesi F, Irrera F, Bologna M, Suppa A, Berardelli A (2015) Smart sensors for the recognition of specific human motion disorders in Parkinson's disease. In: 2015 6th international workshop on advances in sensors and interfaces (IWASI), pp. 131-136. IEEE. https://doi.org/10.1109/IWASI.2015.7184973
- Lukšys D., Jatužis D, Kaladytė-Lokorninienė R, Bunevičiũtė R, Sawicki A, Griškevičius J (2018) Differentiation of gait using principal component analysis and application for Parkinson's disease monitoring. In: 2018 international conference BIOMDLORE, pp. 1-4. IEEE. https://doi.org/10.1109/ BIOMDLORE.2018.8467197
- Maculewicz J, Kofoed LB, Serafin S (2016) A technological review of the instrumented footwear for rehabilitation with a focus on Parkinson's disease patients. Front Neurol 7:1. https://doi.org/10.3389/fneur. 2016.00001
- Manish D, Wadhwani AK, Wadhwani S (2013) Gait based vertical ground reaction force analysis for Parkinson's disease diagnosis using self organizing map. Int J Adv Biol Biomed Res 1(6):624–636
- Marianna C, Pepa L, Verdini F, Ceravolo MG (2016) A smartphone-based architecture to detect and quantify freezing of gait in Parkinson's disease. Gait Posture 50:28–33. https://doi.org/10.1016/j.gaitpost. 2016.08.018
- 93. Masiala S (2017) Detection of freezing of gait in patients with Parkinson's disease using deep recurrent neural networks. Dissertation, Tilburg University.
- Masiala S, Huijbers W, Atzmueller M (2019) Feature-Set-Engineering for Detecting Freezing of Gait in Parkinson's Disease using Deep Recurrent Neural Networks. arXiv preprint arXiv:1909.03428
- MASPARK (2015-2017) Freezing in Parkinson's disease: improving quality of life with an automatic control system- La Marató de TV3-20140431
- Mayo Clinic (2019) Parkinson's Disease (online). https://www.mayoclinic.org/diseases-conditions/ parkinsons-disease/symptoms-causes/syc-20376055. Accessed 25 Dec 2019
- Mazilu S, Hardegger M, Zhu Z, Roggen D, Tröster G, Plotnik M, Hausdorff JM (2012) Online detection of freezing of gait with smartphones and machine learning techniques. In: 2012 6th international conference on pervasive computing Technologies for Healthcare (pervasive health) and workshops, pp. 123-130. IEEE. https://doi.org/10.4108/icst.pervasivehealth.2012.248680
- Mazilu S, Calatroni A, Gazit E, Roggen D, Hausdorff JM, Tröster G (2013). Feature learning for detection and prediction of freezing of gait in Parkinson's disease. In: International workshop on machine learning and data mining in pattern recognition, pp. 144–158. Springer, Berlin, Heidelberg
- Mazilu S, Blanke U, Roggen D, Troster G, Gazit E, Hausdorff JM (2013) Engineers meet clinicians: augmenting Parkinson's disease patients to gather information for gait rehabilitation. In: Proceedings of the 4th Augmented Human International Conference, ACM. pp. 124–127. https://doi.org/10.1145/2459236. 2459257
- Mazilu S, Calatroni A, Gazit E, Mirelman A, Hausdorff JM, Tröster G (2015) Prediction of freezing of gait in Parkinson's from physiological wearables: an exploratory study. IEEE J Biomed Health Inf 19(6):1843– 1854. https://doi.org/10.1109/JBHI.2015.2465134

- Mazilu S, Blanke U, Calatroni A, Gazit E, Hausdorff JM, Tröster G (2016) The role of wrist-mounted inertial sensors in detecting gait freeze episodes in Parkinson's disease. Pervasive Mob Comput 33:1–16. https://doi.org/10.1016/j.pmcj.2015.12.007
- Medeiros L, Almeida H, Dias L, Perkusich M, Fischer R (2016) A gait analysis approach to track parkinson's disease evolution using principal component analysis. In: 2016 IEEE 29th International Symposium on Computer-Based Medical Systems (CBMS), pp. 48–53. IEEE, 2016. https://doi.org/10. 1109/CBMS.2016.14
- 103. Micó-Amigo ME, Kingma I, Faber GS, Kunikoshi A, van Uem JMT, van Lummel RC, Maetzler W, van Dieën JH (2017) Is the assessment of 5 meters of gait with a single body-fixed-sensor enough to recognize idiopathic Parkinson's disease-associated gait? Ann Biomed Eng 45(5):1266–1278. https://doi.org/10. 1007/s10439-017-1794-8
- 104. Mico-Amigo E, Kingma I, Heinzel S, Rispens S, Heger T, Nussbaum S, Van Lummel R, Berg D, Maetzler W, Van Dieen J (2019) Potential markers of progression in idiopathic Parkinson's disease derived from assessment of circular gait with a single body-fixed-sensor: a 5-year longitudinal study. Front Hum Neurosci 13:59. https://doi.org/10.3389/fnhum.2019.00059
- Mileti I, Germanotta M, Alcaro S, Pacilli A, Imbimbo I, Petracca M, Erra C et al. (2017) Gait partitioning methods in Parkinson's disease patients with motor fluctuations: A comparative analysis. In: 2017 IEEE International Symposium on Medical Measurements and Applications (MeMeA), pp. 402–407. https://doi. org/10.1109/MeMeA.2017.7985910
- Mileti I, Germanotta M, Di Sipio E, Imbimbo I, Pacilli A, Erra C, Petracca M et al (2018) Measuring gait quality in parkinson's disease through real-time gait phase recognition. Sensors 18(3):919. https://doi.org/ 10.3390/s18030919
- Minamisawa T, Sawahata H, Takakura K, Yamaguchi T (2012) Characteristics of temporal fluctuation of the vertical ground reaction force during quiet stance in Parkinson's disease. Gait Posture 35(2):308–311. https://doi.org/10.1016/j.gaitpost.2011.09.106
- Mitoma H, Yoneyama M, Orimo S (2010) 24-hour recording of parkinsonian gait using a portable gait rhythmogram. Intern Med 49(22):2401–2408. https://doi.org/10.2169/internalmedicine.49.3511
- Mittra Y, Rustagi V (2018) Classification of Subjects with Parkinson's Disease Using Gait Data Analysis. In: 2018 International Conference on Automation and Computational Engineering (ICACE), pp. 84–89. IEEE, 2018. https://doi.org/10.1109/ICACE.2018.8687022
- Moore ST, Yungher DA, Morris TR, Dilda V, MacDougall HG, Shine JM, Naismith SL, Lewis SJG (2013) Autonomous identification of freezing of gait in Parkinson's disease from lower-body segmental accelerometry. J Neuroeng Rehab 10(1):19
- 111. Muro-De-La-Herran A, Garcia-Zapirain B, Mendez-Zorrilla A (2014) Gait analysis methods: an overview of wearable and non-wearable systems, highlighting clinical applications. Sensors 14(2):3362–3394. https://doi.org/10.3390/s140203362
- Myers PS, McNeely ME, Pickett KA, Duncan RP, Earhart GM (2018) Effects of exercise on gait and motor imagery in people with Parkinson disease and freezing of gait. Parkinsonism Relat Disord 53:89–95. https://doi.org/10.1016/j.parkreldis.2018.05.006
- Nantel J, de Solages C, Bronte-Stewart H (2011) Repetitive stepping in place identifies and measures freezing episodes in subjects with Parkinson's disease. Gait Posture 34(3):329–333. https://doi.org/10. 1016/j.gaitpost.2011.05.020
- 114. Nguyen H, Lebel K, Bogard S, Goubault E, Boissy P, Duval C (2017) Using inertial sensors to automatically detect and segment activities of daily living in people with Parkinson's disease. IEEE Trans Neural Syst Rehab Eng 26(1):197–204. https://doi.org/10.1109/TNSRE.2017.2745418
- 115. Nguyen AN, Roth N, Nooshin Haji Ghassemi R, Hannink J, Seel T, Klucken J, Gassner H, Eskofier BM (2019) Development and clinical validation of inertial sensor-based gait-clustering methods in Parkinson's disease. J Neuroeng Rehab 16(1):77. https://doi.org/10.1186/s12984-019-0548-2
- Nixon MS, Carter JN (2004) Advances in automatic gait recognition. In: Sixth IEEE International Conference on Automatic Face and Gesture Recognition, 2004. Proceedings, pp. 139–144. https://doi. org/10.1109/afgr.2004.1301521
- 117. Okuda S, Takano S, Ueno M, Hara Y, Chida Y, Ikkaku T, Kanda F, Toda T (2016) Gait analysis of patients with Parkinson's disease using a portable triaxial accelerometer. Neurol Clin Neurosci 4(3):93–97. https://doi.org/10.1111/ncn3.12043
- Okuma YH, Mitoma MY (2015) Detection and quantification of freezing of gait and falls in Parkinson's disease patients using a body-worn sensor. J Neurol Sci 357:e281. https://doi.org/10.1016/j.jns.2015.08. 984
- Ornelas-Vences C, Sanchez-Fernandez LP, Sanchez-Perez LA, Garza-Rodriguez A, Villegas-Bastida A (2017) Fuzzy inference model evaluating turn for Parkinson's disease patients. Comput Biol Med 89:379– 388. https://doi.org/10.1016/j.compbiomed.2017.08.026

- Orozco-Arroyave JRJ, Arias-Londoño D, Vargas-Bonilla JF, Gonzalez-Rativa MC, Nöth E (2014) New Spanish speech corpus database for the analysis of people suffering from Parkinson's disease. In LREC 342–347
- Ortells J, Herrero-Ezquerro MT, Mollineda RA (2018) Vision-based gait impairment analysis for aided diagnosis. Med Biol Eng Comput 56(9):1553–1564
- 122. Ossig C, Antonini A, Buhmann C, Classen J, Csoti I, Falkenburger B, Schwarz M, Winkler J, Storch A (2016) Wearable sensor-based objective assessment of motor symptoms in Parkinson's disease. J Neural Transm 123(1):57–64. https://doi.org/10.1007/s00702-015-1439-8
- Ota L, Uchitomi H, Orimo S, Miyake Y (2012). Classification of Parkinson's disease patients' gait variability. In: 2012 IEEE/SICE International Symposium on System Integration (SII), pp. 343–348. https://doi.org/10.1109/SII.2012.6427372
- Oung QW, Basah SN, Muthusamy H, Vijean V, Lee H, Khairunizam W, Bakar SA, Razlan ZM, Ibrahim Z (2018) Objective evaluation of freezing of gait in patients with Parkinson's Disease through machine learning approaches. In: 2018 International Conference on Computational Approach in Smart Systems Design and Applications (ICASSDA), pp. 1–7. https://doi.org/10.1109/ICASSDA.2018.8477606
- Paragliola G, Coronato A (2018) Gait anomaly detection of subjects with Parkinson's disease using a deep time series-based approach. IEEE Access 6:73280–73292. https://doi.org/10.1109/ACCESS.2018.2882245
- Pardoel S, Kofman J, Nantel J, Lemaire ED (2019) Wearable-sensor-based detection and prediction of freezing of gait in Parkinson's disease: a review. Sensors 19(23):5141. https://doi.org/10.3390/s19235141
- Parkinson's Foundation (2018) About Parkinson's Disease (online). https://www.parkinson.org/. Accessed 28 Jan 2020
- Parkinson's Progression Markers Initiative (2018) A landmark Study of Parkinson's Disease (online). https://www.ppmi-info.org/. Accessed 15 Mar 2020
- 129. Peng TD, Ren (2017) Texture classification and visualization of time series of gait dynamics in patients with neuro-degenerative diseases. IEEE Trans Neural Syst Rehab Eng 26(1):188–196. https://doi.org/10. 1109/TNSRE.2017.2732448
- Pepa L, Ciabattoni L, Verdini F, Capecci M, Ceravolo MG (2014) Smartphone based fuzzy logic freezing of gait detection in parkinson's disease. In: 2014 IEEE/ASME 10th International Conference on Mechatronic and Embedded Systems and Applications (MESA), pp. 1–6. IEEE, 2014. https://doi.org/ 10.1109/MESA.2014.6935630
- 131. Pepa L, Andrenelli E, Di Biagio L, Ciabattoni L, Spalazzi L, Ceravolo MG, Capecci M (2018) Reliability of a smartphone-based home monitoring of freezing of gait in subjects with Parkinson's disease. Ann Phys Rehab Med 61:e437–e438. https://doi.org/10.1016/j.rehab.2018.05.1017
- Pereira MP, Gobbi LTB, Almeida QJ (2016) Freezing of gait in Parkinson's disease: evidence of sensory rather than attentional mechanisms through muscle vibration. Parkinsonism Relat Disord 29:78–82. https:// doi.org/10.1016/j.parkreldis.2016.05.021
- Pereira CR, Pereira DR, Weber SAT, Hook C, de Albuquerque VHC, Papa JP (2019) A survey on computer-assisted Parkinson's disease diagnosis. Artif Intell Med 95:48–63. https://doi.org/10.1016/j. artmed.2018.08.007
- 134. Pérez-Toro PA, Vásquez-Correa JC, Arias-Vergara T, Garcia-Ospina N, Orozco-Arroyave JR, Nöth E (2018) A non-linear dynamics approach to classify gait signals of patients with Parkinson's disease. In: Workshop on engineering applications, pp. 268–278. Springer, Cham. https://doi.org/10.1007/978-3-030-00353-1\_24
- Perumal SV, Sankar R (2016) Gait and tremor assessment for patients with Parkinson's disease using wearable sensors. Ict Express 2(4):168–174. https://doi.org/10.1016/j.icte.2016.10.005
- Pham TT, Moore ST, Lewis SJG, Nguyen DN, Dutkiewicz E, Fuglevand AJ, McEwan AL, Leong PHW (2017) Freezing of gait detection in parkinson's disease: a subject-independent detector using anomaly scores. IEEE Trans Biomed Eng 64(11):2719–2728. https://doi.org/10.1109/TBME.2017.2665438
- 137. Pichler D, Fietzek U, Pfister F, Ahmadi A, Achilles F, Abedinpour K, Bötzel K, Ceballos-Baumann A (2017) Long-term detection of motor fluctuations by wrist-worn sensors in Parkinson's disease. Basal Ganglia 8:22. https://doi.org/10.1016/j.baga.2017.02.066
- Piro NE, Piro LK, Kassubek J, Blechschmidt-Trapp RA (2016) Analysis and visualization of 3D motion data for UPDRS rating of patients with Parkinson's disease. Sensors 16(6):930. https://doi.org/10.3390/ s16060930
- Prakesh C, Kumar R, Mittal N (2018) Recent developments in human gait research: parameters, approaches, applications, machine learning techniques and challenges. Artif Intell Rev 49(1):1–40. https://doi.org/10.1007/s10462-016-9514-6
- Prateek GV, Skog I, McNeely ME, Duncan RP, Earhart GM, Nehorai A (2017) Modeling, detecting, and tracking freezing of gait in Parkinson disease using inertial sensors. IEEE Trans Biomed Eng 65(10):2152– 2161. https://doi.org/10.1109/TBME.2017.2785625

- Protokinetics Team (2018) Understanding Phases of the Gait Cycle (online). https://www.protokinetics. com/2018/11/28/understanding-phases-of-the-gait-cycle/. Accessed 12 April 2019
- 142. Punin C, Barzallo B, Huerta M, Bermeo J, Llumiguano C, Soto A, Clotet R (2017) Wireless system for detection of FOG in patients with Parkinson's disease. In: 2017 global medical engineering physics exchanges/Pan American health care exchanges (GMEPE/PAHCE), pp. 1-4. IEEE
- Punin C, Barzallo B, Clotet R, Bermeo A, Bravo M, Bermeo JP, Llumiguano C (2019) A non-invasive medical device for parkinson's patients with episodes of freezing of gait. Sensors 19(3):737. https://doi. org/10.3390/s19030737
- Putri FT, Caesarendra W, Ariyanto M, Pasmanasari ED (2016) Electromyography gait test for Parkinson disease recognition using artificial neural network classification in Indonesia. Jurnal Ilmiah Momentum 12(2). https://doi.org/10.36499/jim.v12i2.1630
- Raccagni C, Gaßner H, Eschlboeck S, Boesch S, Krismer F, Seppi K, Poewe W et al (2018) Sensor-based gait analysis in atypical parkinsonian disorders. Brain Behav 8(6):e00977. https://doi.org/10.1002/brb3.977
- Ravi T, Rajesh R, Sivasankar A (2018) Posture instability of people diagnosed with Parkinson's disease-a review. Int J Eng Technol 7(3–6):119–123
- 147. Raykov YP, Evers LJW, Badawy R, Faber MJ, Bloem BR, Claes K, Little MA (2018) Probabilistic modelling of gait for remote passive monitoring applications. arXiv preprint arXiv:1812.02585
- 148. Rehman RZU, Del Din S, Shi JQ, Galna B, Lord S, Yarnall AJ, Yu G, Rochester L (2019) Comparison of walking protocols and gait assessment Systems for Machine Learning-Based Classification of Parkinson's disease. Sensors 19(24):5363. https://doi.org/10.3390/s19245363
- 149. Ren P, Zhao W, Zhao Z, Bringas-Vega ML, Valdes-Sosa PA, Kendrick KM (2015) Analysis of gait rhythm fluctuations for neurodegenerative diseases by phase synchronization and conditional entropy. IEEE Trans Neural Syst Rehab Eng 24(2):291–299. https://doi.org/10.1109/TNSRE.2015.2477325
- 150. Ren P, Tang S, Fang F, Luo L, Xu L, Bringas-Vega ML, Yao D, Kendrick KM, Valdes-Sosa PA (2016) Gait rhythm fluctuation analysis for neurodegenerative diseases by empirical mode decomposition. IEEE Trans Biomed Eng 64(1):52–60. https://doi.org/10.1109/TBME.2016.2536438
- Rennie L, Dietrichs E, Moe-Nilssen R, Opheim A, Franzén E (2017) The validity of the gait variability index for individuals with mild to moderate Parkinson's disease. Gait Posture 54:311–317. https://doi.org/ 10.1016/j.gaitpost.2017.03.023
- 152. Rennie L, Löfgren N, Moe-Nilssen R, Opheim A, Dietrichs E, Franzén E (2018) The reliability of gait variability measures for individuals with Parkinson's disease and healthy older adults-the effect of gait speed. Gait Posture 62:505–509. https://doi.org/10.1016/j.gaitpost.2018.04.011
- 153. Rewar S (2015) A systematic review on Parkinson's disease (PD). Indian J Res Pharm Biotechnol 3(2):176
- Rizek P, Kumar N, Jog MS (2016) An update on the diagnosis and treatment of Parkinson disease. CMAJ 188(16):1157–1165. https://doi.org/10.1503/cmaj.151179
- 155. Rocchi L, Palmerini L, Weiss A, Herman T, Hausdorff JM (2013) Balance testing with inertial sensors in patients with Parkinson's disease: assessment of motor subtypes. IEEE Trans Neural Syst Rehab Eng 22(5):1064–1071. https://doi.org/10.1109/TNSRE.2013.2292496
- 156. Rodríguez-Martín D, Samà A, Pérez-López C, Cabestany J, Català A, Rodríguez-Molinero A (2015) Posture transition identification on PD patients through a SVM-based technique and a single waist-worn accelerometer. Neurocomputing 164:144–153. https://doi.org/10.1016/j.neucom.2014.09.084
- 157. Rodríguez-Martín D, Samà A, Pérez-López C, Català A, Moreno JM, Arostegui JC, Bayés À et al (2017) Home detection of freezing of gait using support vector machines through a single waist-worn triaxial accelerometer. PloS 12(2). https://doi.org/10.1371/journal.pone.0171764
- 158. Roggendorf J, Chen S, Baudrexel S, Van De Loo S, Seifried C, Hilker R (2012) Arm swing asymmetry in Parkinson's disease measured with ultrasound based motion analysis during treadmill gait. Gait Posture 35(1):116–120. https://doi.org/10.1016/j.gaitpost.2011.08.020
- Rovini E, Maremmani C, Moschetti A, Esposito D, Cavallo F (2018) Comparative motor pre-clinical assessment in Parkinson's disease using supervised machine learning approaches. Ann Biomed Eng 46(12):2057–2068. https://doi.org/10.1007/s10439-018-2104-9
- 160. Saad A (2016) Detection of freezing of gait in Parkinson's disease. Dissertation, Le Havre
- 161. Sakar BE, Erdem Isenkul M, Okan Sakar C, Sertbas A, Gurgen F, Delil S, Apaydin H, Kursun O (2013) Collection and analysis of a Parkinson speech dataset with multiple types of sound recordings. IEEE J Biomed Health Inf 17(4):828–834
- 162. Sama A, Pérez-López C, Rodríguez-Martín D, Català A, Moreno-Aróstegui JM, Cabestany J, de Mingo E, Rodríguez-Molinero A (2017) Estimating bradykinesia severity in Parkinson's disease by analysing gait through a waist-worn sensor. Comput Biol Med 84:114–123. https://doi.org/10.1016/j.compbiomed.2017. 03.020
- 163. Samà A, Rodríguez-Martín D, Pérez-López C, Català A, Alcaine S, Mestre B, Anna P, Cruz Crespo M, Bayés À (2018) Determining the optimal features in freezing of gait detection through a single waist

accelerometer in home environments. Pattern Recogn Lett 105:135-143. https://doi.org/10.1016/j.patrec. 2017.05.009

- Sarbaz Y, Banaie M, Pooyan M, Gharibzadeh S, Towhidkhah F, Jafari A (2012) Modeling the gait of normal and parkinsonian persons for improving the diagnosis. Neurosci Lett 509(2):72–75. https://doi.org/ 10.1016/j.neulet.2011.10.002
- Schlachetzki JCM, Barth J, Marxreiter F, Gossler J, Kohl Z, Reinfelder S, Gassner H et al (2017) Wearable sensors objectively measure gait parameters in Parkinson's disease. PLoS One 12(10). https://doi.org/10. 1371/journal.pone.0183989
- 166. Segovia F, Górriz JM, Ramírez J, Martínez-Murcia FJ, Levin J, Schuberth M, Brendel M et al (2017) Multivariate analysis of 18F-DMFP PET data to assist the diagnosis of parkinsonism. Front Neuroinforma 11:23. https://doi.org/10.3389/fninf.2017.00023
- Seidenberg PH, Beutler AI (2008) The sports medicine resource manual. Elsevier. https://doi.org/10.1016/ B978-1-4160-3197-0.X1000-2
- Sereix JC (2017) Applying deep-learning techniques to detect freezing of gait episodes in Parkinson's disease patients
- 169. Shawn (2019) What is EMG sensor, Myoware and how to use with Arduino (online). https://www. seeedstudio.com/blog/2019/12/27/what-is-emg-sensor-myoware-and-how-to-use-with-arduino/. Accessed 21 Nov 2019
- Shetty S, Rao YS (2016) SVM based machine learning approach to identify Parkinson's disease using gait analysis. In: 2016 International conference on inventive computation technologies (ICICT), vol. 2, pp. 1-5. IEEE. https://doi.org/10.1109/INVENTIVE.2016.7824836
- 171. Shiwani B, Roy S, Kline J, Saint-Hilaire M, Thomas C, Gennert M, De Luca G (2017) Improved detection of gait abnormalities in Parkinson's disease using an IMU sensor-based system. In: Movement Disorders, 32. 111 RIVER ST, HOBOKEN 07030-5774, NJ USA: Wiley
- Shreya B, Rajendra Acharya U, Hagiwara Y, Dadmehr N, Adeli H (2018) Parkinson's disease: cause factors, measurable indicators, and early diagnosis. Comput Biol Med 102:234–241. https://doi.org/10. 1016/j.compbiomed.2018.09.008
- 173. Shrivastava P, Shukla A, Vepakomma P, Bhansali N, Verma K (2017) A survey of nature-inspired algorithms for feature selection to identify Parkinson's disease. Comput Methods Prog Biomed 139: 171–179. https://doi.org/10.1016/j.cmpb.2016.07.029
- Shyam VM, Sankar R (2016) Gait monitoring system for patients with Parkinson's disease using wearable sensors. In: 2016 IEEE Healthcare Innovation Point-Of-Care Technologies Conference (HI-POCT), pp. 21–24. IEEE, 2016. https://doi.org/10.1109/HIC.2016.7797687
- Sijobert B, Denys J, Azevedo Coste C, Geny C (2014) IMU based detection of freezing of gait and festination in Parkinson's disease. In: 2014 IEEE 19th International Functional Electrical Stimulation Society Annual Conference (IFESS), pp. 1–3. https://doi.org/10.1109/IFESS.2014.7036751
- Sneha Baby M, Saji AJ, Sathish Kumar C (2017) Parkinsons disease classification using wavelet transform based feature extraction of gait data. In 2017 International Conference on Circuit, Power and Computing Technologies (ICCPCT), pp. 1–6. IEEE, 2017. https://doi.org/10.1109/ICCPCT.2017.8074230
- 177. Soubra R, Diab MO, Moslem B (2016) Identification of Parkinson's disease by using multichannel Vertical Ground Reaction Force signals. In: 2016 International Conference on Bio-engineering for Smart Technologies (BioSMART), pp. 1–4. IEEE, 2016. https://doi.org/10.1109/BIOSMART.2016. 7835604
- Stamatakis J, Cremers J, Maquet D, Macq B, Garraux G (2011) Gait feature extraction in Parkinson's disease using low-cost accelerometers. In: 2011 annual international conference of the IEEE engineering in medicine and biology society, pp. 7900-7903. IEEE. https://doi.org/10.1109/IEMBS.2011.6091948
- Statista Research Department (2010) Parkinson's disease: projected worldwide increase in prevalence (online). https://www.statista.com/statistics/215459/projected-worldwide-increase-in-prevalence-ofparkinsons-diseas/. Accessed 3 Mar 2020
- Su R-H, Hsu Y-L, Chan L, Lin H, Yang C-C (2014) Assessing abnormal gaits of parkinson's disease patients using a wearable motion detector. Biomed Eng: Appl Basis Commun 26(2):1450031. https://doi. org/10.4015/S1016237214500318
- 181. Tay A, Yen SC, Lee PY, Wang CY, Neo A, Phan SW, Yogaprakash K, Liew S, Au WL (2015) Freezing of gait (FoG) detection for Parkinson disease. In 2015 10th Asian control conference (ASCC), pp. 1-6, IEEE. https://doi.org/10.1109/ASCC.2015.7244608
- Tekscan (n.d.) Why is Gait Analysis Important? (online). https://www.tekscan.com/blog/medical/whygait-analysis-important. Accessed 5 Jan 2020
- 183. The Lancet (2017) Global, regional, and national incidence, prevalence, and years lived with disability for 328 diseases and injuries for 195 countries, 1990–2016: a systematic analysis for the Global Burden of Disease Study 2016. Global Health Metrics 390:1211–1259

- Thompson E, Agada P, Geoffrey Wright W, Reimann H, Jeka J (2017) Spatiotemporal gait changes with use of an arm swing cueing device in people with Parkinson's disease. Gait Posture 58:46–51. https://doi. org/10.1016/j.gaitpost.2017.07.001
- Tien I, Glaser SD, Aminoff MJ (2010) Characterization of gait abnormalities in Parkinson's disease using a wireless inertial sensor system. In 2010 annual international conference of the IEEE engineering in medicine and biology, pp. 3353-3356. IEEE. https://doi.org/10.1109/IEMBS.2010.5627904
- Tonn Niazmand KK, Zhao Y, Fietzek UM, Schroeteler F, Ziegler K, Ceballos-Baumann AO, Lueth TC (2011) Freezing of Gait detection in Parkinson's disease using accelerometer based smart clothes. In: 2011 IEEE Biomedical Circuits and Systems Conference (BioCAS), pp. 201–204. https://doi.org/10.1109/ BioCAS.2011.6107762
- Torvi VG, Bhattacharya A, Chakraborty S (2018) Deep Domain Adaptation to Predict Freezing of Gait in Patients with Parkinson's Disease. In: 2018 17th IEEE International Conference on Machine Learning and Applications (ICMLA), pp. 1001–1006. https://doi.org/10.1109/ICMLA.2018.00163
- Tripoliti EE, Tzallas AT, Tsipouras MG, Rigas G, Bougia P, Leontiou M, Konitsiotis S, Chondrogiorgi M, Tsouli S, Fotiadis DI (2013) Automatic detection of freezing of gait events in patients with Parkinson's disease. Comput Methods Prog Biomed 110(1):12–26. https://doi.org/10.1016/j.cmpb.2012.10.016
- Tzallas AT, Tsipouras MG, Rigas G, Tsalikakis DG, Karvounis EC, Chondrogiorgi M, Psomadellis F et al (2014) PERFORM: a system for monitoring, assessment and management of patients with Parkinson's disease. Sensors 14(11):21329–21357. https://doi.org/10.3390/s141121329
- UCI Machine Learning Respository (2017) Parkinson's Dataset (online). https://archive.ics.uci.edu/ml/ datasets/Parkinsons. Accessed 10 Mar 2020
- Uustal H (n.d.). Normal Gait (online). https://www.medicine.missouri.edu/sites/default/files/Normal-Gaitilovepdf-compressed.pdf. Accessed 21 Sept 2019
- 192. van Wegen EEH, de Goede CJT, Kwakkel G, van Kordelaar J (2018) Sensor assisted self-management in Parkinson's disease: a feasibility study of ambulatory posture detection and feedback to treat stooped posture. Parkinsonism Relat Disord 46:S57–S61. https://doi.org/10.1016/j.parkreldis.2017.07.024
- Verlekar TT, Soares LD, Correia PL (2018) Automatic classification of gait impairments using a markerless 2D video-based system. Sensors 18(9):2743. https://doi.org/10.3390/s18092743
- Wang K, Li B, Gu D, Dai K, Zhou L (2017) A smartphone based system for freezing of gait monitoring for parkinson's disease patients. In: 2017 IEEE 9th International Conference on Communication Software and Networks (ICCSN), pp. 1529–1533. IEEE, 2017. https://doi.org/10.1109/ICCSN.2017.8230364
- Watelain E (2017) Human gait: from clinical gait analysis to diagnosis assistance. Movement Sport Sciences 4:3–4. https://doi.org/10.3917/sm.098.0003
- 196. Weber SAT, dos Santos Filho CA, Shelp AO, Lima Resende LA, Papa JP, Hook C (2014) Classification of handwriting patterns in patients with Parkinson's disease, using a biometric sensor. Glob Adv Res J Med Med Sci:362–366
- Woolf AD (2015) Global burden of osteoarthritis and musculoskeletal diseases. BMC Musculoskelet Disord 16(1):S3. https://doi.org/10.1186/1471-2474-16-S1-S3
- Wu X, Chen X, Duan Y, Xu S, Cheng N, An N (2017) A study on gait-based Parkinson's disease detection using a force sensitive platform. In: 2017 IEEE International Conference on Bioinformatics and Biomedicine (BIBM), pp. 2330–2332. IEEE, 2017.
- Xia Y, Zhang J, Ye Q, Cheng N, Lu Y, Zhang D (2018) Evaluation of deep convolutional neural networks for detection of freezing of gait in Parkinson's disease patients. mBiom Signal Process Control 46:221– 230. https://doi.org/10.1016/j.bspc.2018.07.015
- Xu C, He J, Zhang X, Wang C, Duan S (2018) Template-matching-based detection of freezing of gait using wearable sensors. Procedia Comput Sci 129:21–27. https://doi.org/10.1016/j.procs.2018.03.038
- Yang K, Xiong W-X, Liu F-T, Sun Y-M, Luo S, Ding Z-T, Wu J-J, Wang J (2016) Objective and quantitative assessment of motor function in Parkinson's disease—from the perspective of practical applications. Ann Transl Med 4(5). https://doi.org/10.21037/atm.2016.03.09
- Yang W, Wang A-N, Ai Q, Sun H-J (2017) An adaptive kernel-based weighted extreme learning machine approach for effective detection of Parkinson's disease. Biomed Signal Process Control 38:400–410. https://doi.org/10.1016/j.bspc.2017.06.015
- Yoneyama M, Kurihara Y, Watanabe K, Mitoma H (2013) Accelerometry-based gait analysis and its application to Parkinson's disease assessment—part 1: detection of stride event. IEEE Trans Neural Syst Rehab Eng 22(3):613–622. https://doi.org/10.1109/TNSRE.2013.2260561
- 204. Yoneyama M, Kurihara Y, Watanabe K, Mitoma H (2013) Accelerometry-based gait analysis and its application to parkinson's disease assessment—part 2: a new measure for quantifying walking behavior. IEEE Trans Neural Syst Rehab Eng 21(6):999–1005. https://doi.org/10.1109/TNSRE.2013.2268251

- Yunfeng K, Chen P, Luo X, Wu M, Liao L, Yang S, Rangayyan RM (2017) Measuring signal fluctuations in gait rhythm time series of patients with Parkinson's disease using entropy parameters. Biomed Signal Process Control 31:265–271. https://doi.org/10.1016/j.bspc.2016.08.022
- Zach H, Janssen AM, Snijders AH, Delval A, Ferraye MU, Auff E, Weerdesteyn V, Bloem BR, Nonnekes J (2015) Identifying freezing of gait in Parkinson's disease during freezing provoking tasks using waist-mounted accelerometry. Parkinsonism Relat Disord 21(11):1362–1366. https://doi.org/10.1016/j.parkreldis.2015.09.051
- 207. Zeng W, Yuan C, Wang Q, Liu F, Wang Y (2019) Classification of gait patterns between patients with Parkinson's disease and healthy controls using phase space reconstruction (PSR), empirical mode decomposition (EMD) and neural networks. Neural Netw 111:64–76. https://doi.org/10.1016/j.neunet.2018.12. 012
- Zhang Y, Ogunbona PO, Li W, Munro B, Wallace GG (2013) Pathological gait detection of Parkinson's disease using sparse representation. In: 2013 International Conference on Digital Image Computing: Techniques and Applications (DICTA), pp. 1–8. IEEE, 2013. https://doi.org/10.1109/DICTA.2013. 6691510
- Zhao A, Lin Q, Li J, Dong J, Yu H (2018) A hybrid spatio-temporal model for detection and severity rating of Parkinson's disease from gait data. Neurocomputing 315:1–8. https://doi.org/10.1016/j.neucom.2018. 03.032
- Zhao A, Qi L, Li J, Dong J, Yu H (2018) LSTM for diagnosis of neurodegenerative diseases using gait data. In: Ninth International Conference on Graphic and Image Processing, vol. 10615, p. 106155B. International Society for Optics and Photonics, 2018
- 211. Zijlstra A, Mancini M, Lindemann U, Chiari L, Zijlstra W (2012) Sit-stand and stand-sit trasitions in older adults and patients with Parkinson's disease: event detection based on motion sensors versus force plates. J Neuroeng Rehab 9(1):75. http://www.jneuroengrehab.com/content/9/1/75. Accessed 14 Mar 2020

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Springer Nature or its licensor holds exclusive rights to this article under a publishing agreement with the author(s) or other rightsholder(s); author self-archiving of the accepted manuscript version of this article is solely governed by the terms of such publishing agreement and applicable law.